Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences

Jung Hee (Ginny) Kim

A thesis submitted to AUT University in partial fulfilment of the requirements for the degree of Master of International Hospitality Management (MIHM)

2008

School of Hospitality and Tourism

Primary Supervisor: Dr Jill Poulston

Table of Contents

Atte	station	of Authorship	5
Ack	nowled	gements	5
Ethi	cs appro	oval	5
Abs	tract		6
CHA	APTER	1	8
INT	RODU	CTION	8
1.1.	Bacl	kground	9
1.2.	Obje	ectives of the study	10
1.3.	Rese	earch questions	10
1.4.	Ove	rview of the study	11
CHA	APTER	2	12
LIT	ERATU	RE REVIEW	12
2.1.	Intro	oduction	13
2.2.	Prof	ile of a manager	13
	2.2.1	Demographic background	14
	2.2.2	Education background	17
	2.2.3	The role of job experiences in becoming a general manager	18
	2.2.4	The length of time taken to become hotel managers	
2.3.	Hos	pitality Management Education	
	2.3.1	Background	
	2.3.2	What kind of education is needed for the hospitality industry	20
	2.3.3	Training programmes (Internship or Cooperative Education Programme).	
2.4.	Care	eer expectations and requirements of students and the hotel industry	
	2.4.1	Negative aspects of expectations and assumptions of students	
	2.4.2	Positive aspects of expectations and assumptions of students	
	2.4.3	Different expectations and assumptions of students and employers	
	2.4.4	Different expectations of educators and employers	
2.5.	Sum	mary	
CHA		3	
		DLOGY	
3.1.		oduction	
3.2.		icipants	
· · · ·		Students	
	3.2.2	Employees	
3.3.		earch method	
0.0.		Questionnaire design	
	3.3.2	Surveys process	
	3.3.3	Analysis of data	
3.4.		itation	
3.5.		mary	
		4	
		¬	
4.1.		oduction	
4.1.		ents' survey	
Τ.Δ.		Students' career expectations for the future	+c 46

	4.2.2	Students' awareness of the working environment in the hospitality indu	stry48
	4.2.3	1 7	
	-	ements for employees	
		Hospitality management education and existing gaps between educators	
		spitality industry	
		Value of a hospitality degree to students	
4.3.		bloyees' survey	
		History backgrounds of managerial employees	
		Expectations of managerial employees as graduate hospitality manager	
		ts during their time of employment at the hotel	
		The hospitality industry needs and the requirements for employees	
		Hospitality management education and existing gaps between educators	
		spitality industry	
		Value of a hospitality management degree to employers	
4.4.		nmary	
		5	
		ON	
5.1.		oduction	
5.2.		lents career expectations and managerial employees working history	
	5.2.1.		
		Expectation of first job field/department and first job entry level	
<i>5</i> 2		Expectations of long term careers	
5.3.		lents' awareness of the working environment in the hospitality industry.	
		ds and requirements of the hospitality industry – view points of students	
emp		Needs and requirement for an ampleyee in the hamitality industry	
	5.4.1	Needs and requirement for an employee in the hospitality industry	
	5.4.2	1 1 2	
5.5.		pitality management education and existing gaps between educators and	
		industrygaps between education and existing gaps between educators and	
nosp		Understandings of both parties - students (or educators) and managers i	
		ality industry	
		Internship or Cooperative Education Programme	
5.6.		ue of a Bachelor Degree of Hospitality Management Qualification	
5.0.		Value and benefits of a Bachelor of Hospitality Management Degree	
	5.6.2	Contribution of a degree in hospitality management in the industry	
5.7.		nmary	
		6	
		IONS AND	
		ENDATIONS	
6.1.		clusion	
6.2.		ommendations	
0.2.	6.2.1	Students' expectations and assumptions of their job entry level	
	6.2.2	Working environment in the hospitality industry	
	6.2.3	Needs and requirements of the hospitality industry	
	6.2.4	Value of a Bachelor Degree of Hospitality Management qualification	
6.3.		as identified for further research	
		CES	
		CES	

Appendix A – Survey Questionnaire for students	. 108
Appendix B – Primary Research Questions and Survey Questionnaire for Students	
Appendix C – Survey Questionnaire for employees	
Appendix D – Primary Research Questions and Survey Questionnaire for managerial	
employees in the hotel industry	
Appendix E – Information sheet for survey	
Appendix F – Ethics approval letter from AUTEC	137
Appendix G – Approval survey for employees	139
Appendix H – Reference letter	
Appendix I – Data modification schedule	
Appendix J - Glossary for this study	
Appendix K – Number of employees by RTO Commercial Accommodation Monitor)	
LIST OF TABLES	
Table 1: Profiles of hotel industry managers ages and gender	1.4
Table 1: Profiles of hotel industry managers – ages and gender	
Table 3: Students' sample (Number of enrolled students as at March 2007)	
Table 4: Managerial employees' sample.	
Table 5: Frequency analysis of job positions with working department of employee	
respondents	
Table 6: Cross-tabulation analysis of students' career expectations for the future - I	
Table 7: Frequency analysis of students' career expectations for the future - II	
Table 8: Frequency analysis of students' awareness of the working environment in the industry - I	
Table 9: Cross-tabulation analysis of students' awareness of the working environment	
the industry –II	
Table 10: Frequency analysis of the hospitality industry needs and the requirements for	
potential employees	
Table 11: Cross-tabulation analysis of hospitality management education and existing	
between educators and the industry	
Table 12: Value of a hospitality degree to students – I	
Table 13: Value of a hospitality degree to students- II	
Table 14: Cross-tabulation analysis of history backgrounds of managerial employees	
Table 15: Cross-tabulation analysis of history backgrounds of managerial employees.	
Table 16: Cross-tabulation analysis of qualification of managerial employees with cur	
working field/department	
Table 17: Frequency analysis of expectation employment from graduates	
Table 18: Frequency analysis of human resource needs and the requirements for poten	
employees	
Table 19: Hospitality management education and existing gaps between educators and	
hospitality industry-I	
Table 20: Hospitality management education and existing gaps between educators and	
hospitality industry-II	
Table 21: Cross-tabulation of analysis of value of a hospitality management degree to	
employers - I	
Table 22: Value of a hospitality management degree to employers – II	
14010 22. Tarac of a nospitality management degree to employers 11	07

Attestation of Authorship

"I hereby declare that this submission is my own work and that, to the best of my knowledge and belief, it contains no material previously published or written by another person (except where explicitly defined in the acknowledgements), nor material which to a substantial extent has been submitted for the award of any other degree or diploma of a university or other institution of higher learning".

Acknowledgements

I would like to thank many people who have given me time in order to produce this thesis and finish my master's degree. Firstly, I would like to thank Dr Jill Poulston and Tracy Harkison, my supervisors, Dr Pat Strauss at AUT University and Rodney Wilson, my proof-reader. Secondly, I would like to thank the participants who contributed data because without their participation I would not be able to complete this thesis. Lastly, without the support of my family and friends I know I could not have accomplished this thesis. Many thanks to my husband, Dae Soo Kim, my children, Grace and Richard, and also my friend, Colin McClean, who have given me the never-ending love and support I have needed to complete my master degree.

Ethics approval

Ethical approval was needed from AUT Ethics Committee (AUTEC) as the research involved human participants. Approval was received on 6th August 2007 with Ethics Application Number 07/59 (Refer to Appendix F).

Abstract

The hospitality industry has seen rapid growth through the last three decades. The hospitality industry in New Zealand is still growing steadily. According to the Ministry of Tourism (2008, para. 3), "Visitor expenditure is forecast (from 2008 to 2014) to increase from \$13.73 billion to \$19.05 billion per annum, an increase of 41.4%" (Ministry of tourism, 2008b). Therefore, in response to the needs of the industry, training providers have established a number of hospitality, leisure, travel and tourism degrees (O'Mahony & Sillitoe, 2001). However, previous studies have identified that hospitality management degrees tend to lack importance in the actual world (Harkison, 2004a, 2004b; Steele; 2003). Hospitality management students seem to rate the value of their degrees higher than do employees in the industry (Collins, 2002; Raybould & Wilkins, 2005). Therefore, there are gaps between the expectations and assumptions of students and those of hospitality professionals, which have led to problems in the hospitality industry.

This study has taken three steps to identify: 1) how career expectations, assumptions and requirements differ between undergraduate hospitality management students and management employees in the hospitality industry; 2) The working histories of managerial employees and their perceptions of the industry's needs; 3) Establish the perceived value of hospitality management degrees amongst hospitality management undergraduate students and the hospitality industry. To do so, a comprehensive literature review was undertaken, and questionnaires were collected from 137 undergraduates enrolled in a Bachelor of International Hospitality Management degree at AUT University and 74 managerial employees in the hotel industry. After an analysis of the data, the researcher examined and compared the results of both groups' data with relation to the findings of the literature review.

The following key points emerged from the research: most students expected that they would work in the hospitality industry after completing their course and supposed that their first work field would be in a Food and Beverage department at a management trainee level. The results of the survey show a match between students' long-term career expectations and the working history of employees. Students also seem to understand the environment of the hospitality industry well. These results show that students organise well, planning for their future careers, and these results were similar with those of some previous studies (Brien, 2004; Harkison, 2004b; Jenkins, 2001).

The results of this study identified similarities and differences of perceptions between undergraduate students and managerial employees in the hotel industry. Similarities of students' perceptions and managerial employees' perceptions are;

- Commitment is rated as the most important factor in determining success, while
 personality is rated the most important attribute for an employee in the
 hospitality industry.
- A degree in hospitality management is not recognised as an important qualification by students or employee respondents. Working experience is more important than a bachelor degree of hospitality management for a prospective employee.
- Students and managerial employees believe that the hospitality educators know the industry well, although educators and employees have different understandings of career expectations.
- Both parties believe that internship might help hospitality students' careers in the industry.

Differences of perceptions between undergraduate students and managerial employees are;

- Students believe that knowledge of the industry and experience are the most significant factors for a new employee in the hospitality industry but employees indicate personality.
- Employees believe that using initiative skills are the most important factor for an employee's career development, whereas students answer communication skills.
- Students believe that a hospitality management bachelor degree will contribute more to the hotel business but employees believe that having three years' experience will contribute more.
- Students consider that an employee with a bachelor's degree in hospitality will receive higher starting salaries than someone without, but employees disagree.
- Students suggest that a bachelor of hospitality management degree will affect opportunities for promotion, but employees do not share that opinion.

This study found that even if students and managerial employees share some common opinions, many gaps still exist between them. Therefore, the researcher suggests recommendations relating to students' expectations and assumptions of their job entry level, the working environment in the hospitality industry, the needs of the hospitality industry, and the value of a bachelors degree of hospitality management. Furthermore, the researcher identifies some limitations of this study and areas for further research.

CHAPTER 1 INTRODUCTION

1.1. Background

The hospitality industry is rated to be one of the fastest growing industries in New Zealand. The Ministry of Tourism (2008a, para. 7) in its forecast of visitor nights from 2007 to 2014, forecast an increase of

"15 million or 15.3%, averaging 2.1% increase per annum. The growth will be driven by an additional 12.8 million international visitor nights (up from 46.7 to 60.6 million) and 2.9 million domestic visitor nights (up from 47.2 to 60.0 million) and 2.4 million domestic visitor nights (up from 52.1 to 54.5 million)" (Ministry of Tourism, 2008a).

The forecast suggests an excellent outlook for tourism in New Zealand (Ministry of Tourism, 2008a). This rise in visitor numbers would appear to indicate that more employees will be needed in the hospitality industry, and it seems logical to assume that there will be a need for qualified hospitality employees (Brien, 2004; Ladkin & Riley, 1996; O'Mahony & Sillitoe, 2001). However, the industry does not appear to regard the bachelor of hospitality qualifications very highly. Consequently, students who have studied for these qualifications are often at a disadvantage when competing for employment with peers who have more experience in the field, but hold few, if any, form of tertiary qualifications (e.g. certificate or diploma or bachelor) (Connor & Pollard, 1996; Harkison, 2004b; Jameson & Holden, 2000; Li & Kivela, 1989; Mason, 1995; Petrova & Mason, 2004). The hospitality industry thus appears to value employees with experience in the field more than those with a bachelor of hospitality. This appears to contrast with the expectation of most people that a degree will always give a graduate an advantage in his or her chosen field area (Jameson & Holden, 2000).

The hospitality industry in New Zealand also seems to place more value on employees with experience in the field (Harkison, 2004a; Steele, 2003). Harkison (2004b) notes that the hospitality industry in New Zealand does not consider a hospitality management degree sufficient for graduates to enter management roles without a certain component of work experience. This is supported by Steele (2003)'s findings that only 39 percent of managerial employees had tertiary degrees, 11 percent had tertiary diplomas, while 50 percent had only completed their secondary school education. These studies suggest experience is regarded by hospitality managers as more important than qualifications, and experienced employees are at a greater advantage than their more formally qualified but less experienced peers.

Hospitality graduates expect to be appointed to professional managerial pathways (e.g. management training programmes) in international hotels because they assume their qualifications will afford them greater opportunities for advancement (Brien, 2004). However, because positions such as management training programmes are limited and a certain amount of experience is required, as well as tertiary qualifications, students are likely to be disappointed (Accorhotels, 2007) due to the difficulty of attaining enough industrial experience while the students are studying. Therefore, a gap exists between the expectations and assumptions of hospitality graduates and their employers.

1.2. Objectives of the study

This study aims to explore the career expectations and assumptions of hospitality students. The expectations and assumptions of Bachelor of International Hospitality Management students at AUT University are compared with the perceptions of managerial employees with regard to the career pathways for new hospitality graduates. Furthermore, this study seeks to identify the current value of a Bachelor of Hospitality Management degree for an individual's intent on a career in hotel management and for managerial employees. The positions of the managerial employees who participated in this study ranged from department supervisors (low management level) to senior-level managers. The participants were employed at six hotels in New Zealand at a large international hotel chain: two from Auckland, one from Rotorua, one from Wellington and two from the South Island.

1.3. Research questions

This study aims to answer the following questions:

- What are the expectations and assumptions of hospitality management students concerning their future working careers?
- What are the working histories of managerial employees and their perceptions of the needs in the hospitality industry?
- What is the value of hospitality degrees to hospitality management students and their employers?

1.4. Overview of the study

Chapter 2 contains a review of the literature relevant to the study, and then chapter 3 discusses the methodology that was used for data collection and analysis. Chapter 4 presents the findings from the data analysis regarding career expectations, assumptions and the requirements of undergraduates of hospitality management and managerial employees in the hotel industry. Chapter 5 incorporates a discussion of these findings. Lastly, chapter 6 presents the conclusions and highlights recommendations arising from the research.

CHAPTER 2 LITERATURE REVIEW

2.1. Introduction

Most undergraduates in a Bachelor of Hospitality Management course expect to be a senior management level employee in the near future (Brien, 2004; Harkison, 2004a). However, the hospitality industry has been changing rapidly as a result of changing economies, societies, and fast developing technologies (Baum, Amoah, & Spivack, 1997). To cope with these changes, employees need various skills and abilities. The industry will benefit from increased levels of formal vocational education and training (O'Mahony & Sillitoe, 2001). Hospitality tertiary educations provide students with the skill sets to succeed in this fast changing industry. Especially undergraduates studying the bachelor of hospitality management programme need to comprehend the contemporary environment in the hospitality industry, and an understanding of the histories of senior managers may assist students in the future development of their own careers. This chapter provides a review of the relevant literature and, in particular, provides profiles of managers in the hospitality industry, a background on hospitality management courses and training programmes, and identifies career expectations and assumptions of hospitality management students and managerial employees in the hospitality industry.

2.2. Profile of a manager

Previous studies have explored the profiles of managerial employees in the hospitality industry (e.g. Betty & Stella, 1998; Harper, Brown, & Irvine, 2005; Ladkin, 1999, 2000; Ladkin & Juwaheer, 2000; Ladkin & Riley, 1996; Nebel, Braunlich, & Zhang, 1994; Riley & Turam, 1989; Steele, 2003). These studies give insight into the background, education and work experience of managers. Hospitality organisations have different organisational structures, but managerial employee groups are often divided into three levels: low management, mid-management and senior management. The low management levels include department supervisors and department assistant managers who assist and help the front line employees to perform their job or task, solve problems, and report to their department managers. The mid-management levels include department managers or department heads that also assist the operational employees and low managerial employees to perform their tasks, while solving higher lever problems and reporting to the top managers. The top management level includes General

Managers (GMs) and regional managers who establish the organisation's purpose and overall direction (Steele, 2003; Wood, Zeffane, Schermerhorn, Hunt, & Osborn, 2001).

2.2.1 Demographic background

Age and gender

The findings from studies relating to profiles of hotel managers are summarised briefly in a table. This table offers an insight into managers' backgrounds and may act as a reference tool in this study.

Table 1: Profiles of hotel industry managers – ages and gender

Authors	Countries	No.	Job Position	Average	Gender		
		sample		Age	Males	Females	
Nebel, Braunlich			Food				
and Zhang			Beverage				
(1994)	America	136	Director	36	96.2%	3.8%	
Nebel, Lee and							
Vidakovic							
(1995)	America	114	Hotel GM	43	92.1%	7.9%	
Ladkin (2000)	United						
	Kingdom	284	Hotel GM	40	87%	13%	
Ladkin and							
Juwaheer (2000)	Mauritania	16	Hotel GM	42	100%	0%	
Ladkin (2002)	Australia	180	Hotel GM	41	80.6%	19.4%	
			Hotel GM	49	73%	17%	
			Department				
Steele (2003)	New		Head	36	56%	44%	
	Zealand	172	Department				
			Manager	31	35%	65%	
			Department		_		
			Supervisor	35	17%	83%	

Nebel, Braunlich and Zhang (1994) found the average age of Food and Beverage Directors (mid-management level) in American hotels to be 36.1 years and 96.2 percent were male. This study was distributed to all Food and Beverage Directors (n=136) of 4 and 5 star USA hotels (all hotels were listed as four and five-star, according to the 1992 Mobil Travel Guide) to gather an appreciation of their unique characteristics, with regard to demographics, education, or experience.

An American study by Nebel, Lee and Vidakovic (1995) explored the career paths of 114 GMs in the United States of America (USA). The study included GMs of midrange (e.g. Holiday Inns), up-scale (e.g. Hyatts, Marriotts and Hiltons) and luxury hotels (Ritz-Carltons and Four Seasons) of 150 to 1300 rooms. The average size of

mid-range hotels was 277 rooms, up-scale hotels 525 rooms and luxury hotels 470 rooms. They concluded that GMs were young (averaging 43 years) and 92.1 percent were male.

Ladkin (2000) examined the predominance of "vocational education and food and beverage experience in the development of a hotel manager's career" (p.226). This study included 284 hotel managers in the United Kingdom (UK), and found the average age of UK hotel GMs was 40 years, and males occupied 87 percent of those managers.

Ladkin and Juwaheer (2000) explored "the career paths of hotel managers in Mauritius" (p.119). They found the average age of the GMs to be 41.6 years old, and most of the managers were male. The Mauritius study included only 16 respondents, of whom 12 were GMs. The sample size was smaller given the constraints associated with Mauritius being a small island with a limited number of potential participants (Ladkin & Juwaheer, 2000).

Ladkin (2002) also explored "the career paths of hotel GMs in Australia" (p.379). The study based on a sample of 180 hotel managers. This study found the average age of the managers to be 40.6 years and males occupied 80.6 percent of the positions.

A New Zealand study (Steele, 2003) investigated the career paths of three levels of managements (low management, mid-management and top management) in the hotel industry in Auckland. Steele (2003) found the average age of GMs of hotels in Auckland to be 48.7 years and 83 percent of general manager positions were occupied by men. The mean age of mid-managers was 35.6 years and 56 percent were male. The mean age of low managers was 30.9 years and females filled 65 percent of these positions. Supervisors had a mean age of 34.7 and only 17% were male (Steele, 2003).

The studies of Nebel et al. (1995), Ladkin (2000), Ladkin and Juwaheer (2000), and Ladkin (2002) showed similar results. GMs were in their early 40s and predominantly male. However, there are differences in New Zealand from these results. Steele (2003) found the average age of GMs was higher (49 years) than those in other studies. He also found that managers in low to supervisory positions tended to be younger and female. The average age of mid-management level employees in New Zealand was similar to the American study by Nebel et al. (1994), but there were more female managers (44%)

than in the American study. The American study was conducted more than 13 years earlier and may no longer give an accurate account of the hotel industry in the U.S.A. Therefore, it may be unwise to compare the New Zealand study with the study by Nebel et al. However, GMs in New Zealand appear to take a longer time to reach a GM position in a hotel.

In the New Zealand hospitality industry, female managers tend to achieve more promotions than male managers in low management and supervisory positions (Steele, 2003). However, female managers are less likely to obtain senior management positions, particularly a GM position (Ladkin, 2000, 2002; Ladkin & Juwaheer, 2000; Li& Leung, 2001; Li, Tse, & Xie, 2007; Steele, 2003). A number of studies (Brownell, 1994; Li & Leung, 2001; Nebel, Lee, & Vidakovic, 1995) investigated why there are so few female GMs in the hotel industry, despite the high percentage of female hotel management and catering graduates. The most common reason, according to the aforementioned studies, was a lack of experience in a Food and Beverage (F&B) department. Many women tended to work in housekeeping, and sales and marketing rather than in an F&B department. Female employees may have responsibilities associated with their roles as wives and/or mothers that may make it difficult for them to work in F&B departments. In addition, women may pass up promotional opportunities that conflict with their other roles (Li & Leung, 2001).

Furthermore, Li and Leung (2001) found that women in Asian countries tend to find the late work hours, typical of hotel occupations, difficult to manage. These hours are not congruent with strong social lives and may conflict with domestic duties. This problem also exists in Western and African nations, as suggested in previous studies (Table 1). Therefore, the lack of female GMs is a phenomenon existing across the globe. However, the Australian (Ladkin, 2002) and New Zealand studies (Steele, 2003) identified that more female managers were working in the hotel industry in Australia and New Zealand than in other parts of the world. This suggests that female managers in Australia and New Zealand have a greater opportunity to become a GM than exists in other countries.

2.2.2 Education background

Education is an important component of a hospitality manager's career history. The studies in table 1 also explored hotel managers' educational backgrounds. 83.5 percent of GMs in British hotels had gained specific hotel and catering qualifications and 16.5 percent had a general or non-vocational qualification. Of the 83.5 percent of GMs with hotel and catering qualifications, 31.8 percent had three years' tertiary levels of full-time education, 24.9 percent had two years' full-time, and 18.1 percent had degree or higher level qualifications (Ladkin, 2000). Nebel et al. (1995) found that 46.5 percent of GMs in American hotels had a Bachelor's degree, 14 percent had a Master's degree, and 6.2 percent had only a high school qualification. The studies of Australia and New Zealand GMs found that high management level managers tend to have varying qualifications. An Australian study by Ladkin (2002) found that 6.1 percent of the sample had a postgraduate degree, 7.8 percent had gained a degree from full-time study, 1.7 percent gained a degree level from part-time study, 7.8 percent had an advanced diploma, 13.3 percent a diploma, 7.8 percent a certificate, and 0.6 percent a culinary course or apprenticeship. Other qualifications were held by 2.7 percent of the sample.

A New Zealand study (Steele, 2003) found that 39 percent of GMs had a bachelor's degree or higher and 11 percent had a tertiary diploma. Of those with university qualifications, half had degrees in hospitality and/or tourism, and half held a business degree. Half of the GMs had no qualifications above High School. Mid-managers with diplomas and degrees equalled 42 percent, while 16 percent had obtained tertiary certificates specific to the industry, and 37 percent had only completed high school. This study also found that 43 percent of low managers held diplomas and degrees, including a third with hospitality/tourism specialist qualifications. Of this group, 52 percent had only completed secondary school. This study showed that the low manager groups tend to be higher qualified than their more senior colleagues, although 67 percent of supervisors had no tertiary qualifications.

Ladkin and Juwaheer (2000), and Ladkin and Riley (1996) found that a university education did not seem to affect the rate of promotion into a hotel GM position. However, Harper et al. (2005) found that 64 percent of GM respondents in their study agreed that "vocational qualifications support a 'fast-track' progression to the GM role"

(p.56). Harper et al. (2005) examined "the role of formal qualifications in the career development of contemporary hotel GMs in Scotland" (p.52). Ladkin (2000) also stressed that

"A higher level of education appears to ensure entry into the industry at the assistant manager level, whereas those with lower levels of education begin at the supervisor or department head grades. This is clearly where education is of benefit" (p. 231).

Given the variances between the findings of GMs' background education and the conflicting conclusions drawn in the literature, it is difficult to identify whether a university education affects the rate of promotion or not.

New Zealand GMs have lower levels of education than in other countries. However, the history of hospitality education in New Zealand is short in comparison to that in other parts of the world, which means New Zealand GMs have had fewer opportunities to pursue formal education.

2.2.3 The role of job experiences in becoming a general manager

In order to become a GM in a hotel, prospective senior managers may require considerable operational and managerial experience, including experience in certain specialist roles, such as a department head or assistant managers in the hotel industry. In the New Zealand study, it appears that top managers in Auckland hotels moved through supervisor and low management, then on to mid-management, before obtaining their first GM position (Steele, 2003). This experience was seen as significant training for one's progression towards GM.

Many studies (e.g. Harper et al. 2005; Ladkin, 2000; Ladkin & Riley, 1994, 1995, 1996; Nebel et al, 1994; Steele, 2003) have identified F&B managerial experience as the main career route to a GM position. Ladkin and Riley (1996) found that 49 percent of GMs in U.K. hotels had F&B managerial experience and 60 percent of GMs in Mauritius were from a F&B background (Ladkin & Juwaheer, 2000). An American study (Nebel, Lee, & Vidakovic, 1995) found that most GMs maintain a narrow career path before reaching a GM position. 87.5 percent of GM respondents spent their career in one department before becoming a GM. Three quarters of GMs had a pathway to their positions through two operational departments, 'F&B' and 'Rooms Division (Front

office)'. Front office work experience was ranked second as a career route for reaching a hotel GM position by the above studies. A minority of GMs had come from sales and marketing, accounting and finance, and housekeeping roles (Ladkin, 2002; Nebel, Lee, & Vidakovic, 1995).

Wagner (1993) explored the question of why many hotels were not aware of the importance of sales and marketing, accounting and finance work experience. Wagner found most GMs come from operational backgrounds, such as F&B or Front Office. Therefore, GMs tended to lack sales and marketing skills, although sales and marketing skills are vital for the success of a hotel (Wagner, 1993).

2.2.4 The length of time taken to become hotel managers

It is significant for individuals to consider the length of time it takes to become a hotel GM for their own career planning (Ladkin, 2002). Therefore, many studies (Harper, Brown, & Irvine, 2005; Ladkin, 2002; Ladkin & Juwaheer, 2000; Ladkin & Riley, 1996) examined 'the length of time taken to become a hotel GM'. These studies used either a 'standardised measure' (from the age of 18 until the first GM position), or an 'un-standardised measure' (based on the age the subject first entered the industry after education).

Table 2: The length of time taken to become a hotel GM

Authors	Countries	Average	Standardised	Un-standardised
		years	measure	measure
Nebel et al. (1995)	America	8.9		
Ladkin & Riley (1996)	United		11.19	9.5
•	Kingdom			
Ladkin & Juwaheer (2000)	Mauritania		13.1	16.1
Ladkin (2002)	Australia		14.1	12.6
Steele (2003)	New	11.3		
, ,	Zealand			
Harper et al. (2005)	Scotland		11.10	9.2

The results of these studies (Harper, Brown, & Irvine, 2005; Ladkin, 2002; Ladkin & Juwaheer, 2000; Ladkin & Riley, 1996) demonstrated that the unqualified managers achieved managerial positions at a younger age than qualified managers did. The qualified managers spent more time studying their course. Therefore, they entered the industry later than an unqualified manager did; hence, there is an age difference when

they started their general manager position. However, Harper et al (2005) found that qualified managers tended to be promoted faster than unqualified managers, which suggests that college or university qualifications might affect the 'fast tracking' of an employee.

2.3. Hospitality Management Education

2.3.1 Background

The hospitality industry has changed rapidly as a result of changing economies, societies and fast developing technologies during the past few decades (Baum, Amoah, & Spivack, 1997). The fast changing hospitality industry needs a new variety of managers. These managers need new skills to handle the complexities of a changing corporate environment. Therefore, hospitality educations are becoming more important to hospitality students and the hospitality industry. Hospitality management courses encourage students to think more strategically and to understand how multinational hospitality companies can best be managed. In New Zealand, this is achieved by instilling an awareness and sensitivity to cultural diversity through the learning of other cultures; the development of appropriate transferable and personal skills through personal development modules (New Zealand Education guide, 2007). O'Mahony and Sillitoe (2001) stress that tertiary hospitality educations might make it possible to operate more effectively, achieve promotion, and change job roles.

2.3.2 What kind of education is needed for the hospitality industry

Employees are quite often required to have specific skills and abilities. Spivack (1997) stated that there are 'skill development issues' related to changes occurring within the tourism and hospitality industry. Christou (1999) highlighted that skills should be given priority when recruiting new staff. Many authors (Damitio, 1988; Damitio & Schmidgall, 1993; Hsu & Gregory, 1995; Knutson & Patton, 1992; Ladkin, 1999; Ley, 1980; Riley & Turam, 1989) pointed out that specific skills would be very important for developing a career because hospitality is becoming more globalised. The need for multicultural abilities and skills is more critical for employees working in the industry (Sigala, 2001).

Baum (1990), Finegold, Wagner, and Mason (2000) and Jauhari (2006) stressed that hotel managers should have multiple technical skills. For example, a F&B manager has to make a decision buying stocks, storing, costing, and processing, to serving food to customers (Riley, 2005). Using multiple technical skills allows an F&B manager to control and manage their department, and this requirement remains the same in all situations, regardless of size or quality, be it a grand hotel or formal restaurant. A F&B Manager needs multiple skills in areas including accounting, human relations, marketing, customer service and communication in order to effectively manage a department (Riley, 2005). These skills are required by all department managers if they are to work more effectively and efficiently. Hence, technically multi-skilled staff will ultimately offer greater value to the company.

Jauhari (2006) also stated that multi-skilling allows a person to manage a large number of tasks at the same time. Multi-skilling may affect how employees approach a wider arrangement of employment opportunities. Furthermore, multi-skilling can increase an employee's understanding of the wider processes involved in different kinds of functional activity (Finegold, Wagner, & Mason, 2000). Baum (1990) stated, hospitality graduates needs to have multiple skills to handle the rapidly changing working environment of the hospitality industry. Multi-skilled employees will be more enterprising and adaptable.

Ladkin and Juwaheer (2000) argued that effective communication skills are fundamental for career development in the hospitality industry. Communication skills are argued to be a significant skill by many (e.g. Ruddy, 1990; Achorn, 1992; Knutson & Patton, 1992; Stutts, 1995). For example, Achorn (1992) noted that in order to succeed, GMs in the Sheraton hotel group needed a good education and good communication skills. Knutson and Patton (1992) found that hospitality students believed that effective communication skills were especially important for GMs. However, other skills are also stressed, for example, managerial accounting skills (Damitio, 1988; Damitio & Schmidgall, 1991), and technical, leadership and interpersonal skills (Kay & Russette, 2000).

It appears that multi-technical skills, good communication skills, leadership skills and financial skills are necessary to operate effectively in the hospitality industry. Knowing the skills required by the hospitality industry is vitally important to hospitality students

and education providers. If educators know what the industry wants and requires of a new employee, they can focus on changing their curricula to support the industry's expectations and needs and improve the satisfaction of hospitality students.

2.3.3 Training programmes (Internship or Cooperative Education Programme)

For recruitment into management levels in the hospitality industry, the industry tends to favour applicants that have work experience. Many managers in the hospitality industry believe that practical experience is more valuable than a degree qualification for entering into the management levels in the hospitality industry (Harkison, 2004a). However, hospitality graduates argue that they already have enough practical experience because most graduates complete a period of internship or a Cooperative Education Programme (CEP) within their degree. Raybould and Wilkins (2005) also stated that graduates believed that they have already carried-out an 'apprenticeship' through completion of university study, as most hospitality degree programmes are often combined with practical work experience.

Advantages of internship or Cooperative Education Programme (CEP)

Undergraduate students of hospitality management may gain industry work experience through CEP or an internship programme. These programmes are academic programmes that encourage students to apply theory in work settings and gain greater work competencies (Harkison, 2004a). All hospitality degrees in New Zealand have a CEP or internship component, which students complete as part of the normal academic curriculum (Harkison, 2004a). These work experiences tend to offer many benefits to students, education providers, and companies. Many authors have stated the benefits of CEP and internship programmes (Harkison, 2004a; Jauhari, 2006; Posey, Carlisle, & Smellie, 1988; Ricci, 2005). CEP and internship programme experience is seen as a cornerstone for students when they are employed in their first job in the hospitality industry (Harkison, 2004a).

Harper et al. (2005) found that 75 percent of qualified managers in Scotland had undertaken a 'structured work experience' component within the structure of the course.

They believed that structured work experience was a strongly encouraged method of developing their technical and business skills. Furthermore, students might be able to evaluate their professional growth based on the results of their work experience (Harkison, 2004a; Jauhari, 2006; Posey, Carlisle, & Smellie, 1988; Ricci, 2005). The authors argue that if organisations recognise students' needs and wants and support their enquiries during a CEP or internship, students will more effectively and efficiently learn and work. As a result, hospitality companies will also benefit. Hospitality companies may be better able to hold their employees and encourage high performance. Furthermore, they will be able to use the materials the intern has developed (Posey, Carlisle, & Smellie, 1988).

Students' CEP and internship programmes also benefit education providers. Through these programmes, education providers can modify courses to incorporate a suitable training method used by the industry. Posey et al. (1988) stated, "A real opportunity for education providers is that they learn how to better prepare future graduates for careers in business and industry" (p.64).

Challenges of internship or Cooperative Education Programme

However, some studies have different opinions of the internship programme and CEP (Jauhari, 2006; Waryszak, 1999). According to Purcell and Quinn (1995 as cited in Waryszak, 1999), graduates had difficulty in finding opportunities to develop managerial skills during their internship programme or CEP placements. For example, many students recognise that they work just like front line employees during the whole period of their CEP or internship programme. Students might feel that organisations treat students as a source of cheap labour. Students expect that a CEP or internship offer opportunities to participate in meetings with managers and the opportunity to learn more specific management skills. Therefore, students might be unsatisfied with their experience of CEP or internship. Experience from the CEP or internship will be a significant factor for students in making decisions about whether they want to work in the hospitality industry or not. If the work environment does not meet student expectations, students might regret entering the hospitality industry after graduation, and use their CEP or internship experience to find positions in alternative sectors (Waryszak, 1999).

On the other hand, if organisations give students the opportunity to experience several departments during their internship or CEP programme, they will concentrate more on their study and they will be better prepared for higher-level positions. Therefore, the CEP and internship experiences will be of more value to students and organisations if each understands their counterpart's expectations. However, education providers offer different training periods. Some education providers offer longer training periods than others do, giving students greater learning opportunities and valuable experience (Harkison, 2004a).

2.4. Career expectations and requirements of students and the hotel industry

What does the hospitality industry expect and require of the bachelor of hospitality management students? Moreover, what do students expect of their working career in the hospitality industry? Many authors explored and identified different viewpoints on the career expectations of students and the hospitality industry. The different perceptions between education providers and the industry was recognised to be problematical (Collins, 2002; Harkison, 2004b; Li & Kivela, 1989; Petrova & Mason, 2004; Raybould & Wilkins, 2005).

2.4.1 Negative aspects of expectations and assumptions of students

Some early studies identified that many students recognised the negative characteristics (poorly paid, working unsociable days and times, long hours and multi-skill with shift work) of the industry after starting work in the hospitality industry (Barron & Maxwell, 1993; Johns & McKechnie, 1995). Barron and Maxwell (1993) found that most students in Scottish higher education institutions believed that the hospitality industry offered career opportunities, but students in the final year had an overall negative perception of the industry. Furthermore, Johns and McKechnie (1995) found that over 50 percent of all hospitality graduates at a hotel school in Norwich pursued careers within the hospitality industry. These meant that they recognised negative working environments and were disappointed with the hospitality industry when they had work experience (Barron and Maxwell, 1993).

Students of hospitality management tended to have high ambitions for their future working career when they began their studies, but their ambitions changed after recognising the actual circumstances of the industry. As a result, some students tended to change their goal, which has serious implications for students, education providers, and the hospitality industry. For example, if many people view the working environment of the hospitality industry negatively, then potential students may choose not to study hospitality and education providers will find it difficult to recruit students and provide high quality courses. Furthermore, the hospitality industry will find it difficult to select employees who have good quality qualifications, which also will affect customers. As a result, customers may be dissatisfied with the hospitality industry, and the industry may suffer financially, which may influence suppliers.

2.4.2 Positive aspects of expectations and assumptions of students

Jenkins (2001), Harkison (2004b), and Brien (2004) identified undergraduate students who have positive career expectations. Many graduates start working in the hospitality industry as front line employees. However, students believe that hospitality degrees will make a difference to their careers create greater professionalism in the industry (Brien, 2004; Harkison, 2004a). This belief seems to be held by hospitality management students worldwide. Furthermore, undergraduate students tend to believe that they will find a good job in a big hotel chain after completing their degree course (Brien, 2004; Jenkins, 2001). In an Anglo-Dutch study (Jenkins, 2001) on students' future perceptions, 77.5 percent of the respondents, who were studying in a hospitality degree course in the UK and the Netherlands, believed that after graduating from their degree course they would find a job in international hotel chains, such as Starwood, Hilton, Hyatt, Marriott, Intercontinental, and Accor.

Brien (2004) also stated that degree graduates of hospitality management in New Zealand want to "seek and gain positions in international hotels that provide a professional managerial career path option and associated benefits" (para. 7). Graduates in New Zealand have the same expectations as identified in the results of Jenkins's (2001) study. These studies found that graduates have positive expectations for their career paths, which are similar to the expectations of students of hospitality management around the world.

The large corporations tend to want to employ graduates because employers want employees who are bright in personality as well as intelligent. Big corporate hospitality companies have also developed broad management training programmes to foster the best employees (Accorhotels, 2007; Brien, 2004). For example, Langham Hotels, Hilton Hotels, and Accor Hotels provide customised programmes for graduate students. Langham Hotel Hong Kong has a partnership programme with local universities (Tang, 2006). The Hilton Hotel offers an 'Elevator Programme' and Accor hotels provide a 'Graduate Management Traineeship Programme'. These programmes offer different experiences, which include all the departments of a hotel. Trainees work as full time employees during their training (Accorhotels, 2007). After finishing the programme, a trainee will progress towards a GM position at a faster pace. Brien (2004) stated, "Graduates are expected to undertake ongoing professional development related to their careers, which may well include some technical training" (para. 7). However, these opportunities are limited to graduates, although the opportunities benefit both students and organisations.

Most undergraduate students might have career expectations of their future. Jenkins (2001) also found that 70 percent of students expected to be employed as a GM or corporate manager 10 years after graduating, and 65 percent of students expected to be a department manager five years after graduating (Jenkins, 2001). The results of Jenkins' study (2001) were very close to the results of previous studies by Ladkin and Juwaheer (2000) and Harper et al. (2005). Ladkin and Juwaheer (2002) found the length of time to reach GM roles ranges from 9 to 15 years. Furthermore, Harper et al. (2005) identified that GMs in Scotland spent an average of 9 years and 2 months to be a GM (Table 2). These results showed that students' expectations of career paths and real industry employees' career paths were very close. Harkison (2004b) noted that students were realistic in terms of the management goals they could achieve following their graduation. Students tend to have a plan to apply for middle management positions, such as in the supervisory and trainee areas, after graduation (Harkison, 2004b). In order to achieve their goals, they have to have a precise career plan, which is very important for success in their career development (Harkison, 2004b). However, Jenkins (2001) also found that no students wanted their own business five years after graduating, and only seven percent of students expected to own a business ten years after graduating, which means not many students considered starting their own business after completing their course.

2.4.3 Different expectations and assumptions of students and employers

Studies by Collins (2002), and Raybould, and Wilkins (2005) focused on the career expectations of hospitality management students and hospitality managers. These two studies found that undergraduates have high expectations of their career in hospitality industry. The authors identified some managers of the hospitality industry who tended to disregard the Bachelor of Hospitality Management degree. Collins (2002) stressed that graduates in Turkey expected to enter a management position after graduating from their course, but employers thought that graduates were not suitable to enter a management position. Some managers considered graduates as having a 'know-it-all' patronising attitude, but that they lacked resilience when coping with the hectic industry environment (Collins, 2002). The studies of Collins, and Raybould, and Wilkins found that students and the industry had different perceptions of career expectations and each tended to insist their perceptions were correct. Students seemed to believe in the value of a degree education and were confident in their career path expectations.

The hospitality industry generally tends to place a low value on degree students' education and capabilities. Purcell and Quinn (1996, as cited in Raybould and Wilkins 2005) stated the hospitality industry tends to ignore students' formal qualifications because students hold unrealistic expectations of the responsibilities and skills they will be expected to show. Therefore, the industry believes that "students are over qualified but under experienced for even entry level management positions" (Raybould and Wilkins, 2005. p. 211). As a result, the industry wants new employees or trainees to gain operational experience for the first 12-18 months, although high achievement students are recruited straight into management training programmes (Raybould & Wilkins, 2005).

The hospitality employers seem to consider work experience as more important than a degree in hospitality management when they select a new employee. This is reflected in the working careers of managerial employees in the New Zealand hospitality industry. Harkison (2004a) and Steele (2003) identified that many unqualified employees still work in management levels in the hospitality industry. They also found that many managers do not have a degree that is relevant to hospitality management. Furthermore, many managers in the hospitality industry in New Zealand also believe that practical experience is more valuable than a hospitality degree and the hospitality degree

qualification is not recognised as an important qualification by the hospitality industry in New Zealand (Harkison, 2004b). However, the industry believes the hospitality degree qualification is needed for entering management levels or for specialist jobs, such as sales and marketing, and accounting (Harkison, 2004a).

According to Spivack (1997), employees with tertiary education in tourism and hospitality tend to out-perform colleagues who lack the same level of education. Ladkin (2000), Harper et al. (2005) and Jauhari (2006) recommended that a person who is looking for a job at a management level in the hospitality industry should undertake a formal qualification or vocational course to enhance career development. Jauhari (2006) also stated that unqualified hospitality managers will have limited exposure and the majority of their learning will occur through trial and error in the workplace. Qualified hospitality managers may reduce training periods and learn faster because of their academic background, which will be beneficial for both students and the industry.

2.4.4 Different expectations of educators and employers

There are gaps between the expectations of educators and the hospitality industry. Some studies (Harkison, 2004b; Li & Kivela, 1989; Petrova & Mason, 2004; Ricci, 2005) showed that the hospitality industry does not know what a hospitality degree consists of, and many educators tend not to understand what the hospitality industry wants from new employees. Hospitality management courses are often designed by academic administrators and educators in order to prepare students for manager positions in the future (Jayawardena, 2001; Ricci, 2005) although in fact the education providers and industry have different viewpoints and do not understand each other's perceptions.

Stutts (1995) and Harris and Zhao (2004) stated some ways to reduce the gaps between industrialists and educators. They need a continuing discussion concerning curriculum, students, facilities, and faculty (Stutts, 1995). Harris and Zhao (2004) also stated that education providers and industry executives should consider job swapping which will benefit both sides. For example, a hotel manager would work at a university as a lecturer and a lecturer would work at a hotel as a manager in a certain department. A hotel manager and a lecturer would have time to address issues of concern. Harris and Zhao (2004) stressed that education providers may benefit from job swapping, to

"...include updated information for class lectures, skill updates useful in laboratory settings, professional contacts for networking, expanding research opportunities, curriculum updates, and the opportunity to change the learning environment" (p. 433).

However, educators and hospitality employers need to reach a compromise, as many valuable graduates are disillusioned about their careers. They spent a lot of money and time studying but were not satisfied with their career choice. If they leave the industry, the hospitality industry in New Zealand will suffer, as graduates will search for better prospects overseas, as has occurred in other industries. Stutts (1995) stated that the hospitality industrialists should share their changing needs, expectations, and priorities with the educator. Furthermore, educators need to balance the instruction of primary principles with the results from highly developed, marketplace research (Stutts, 1995).

2.5. Summary

This chapter started by investigating the profile of managers and hospitality management education, and highlighted the career expectations, assumptions and requirements of students and managerial employees in the hotel industry. Previous hospitality studies identified that senior management employees in the hospitality industry were mostly educated young males. The literature was reviewed in order to offer a profile of the working history of hospitality managerial employees. The literature review was then compared with the findings of students' expectations and assumptions and the working history of managerial employees in this study.

The literature is varied in its assessment of hospitality training programmes (internship and cooperative education programme) for undergraduates of hospitality management. In the literature, it is argued that these programmes offer many advantages to students, educators and the industry, but they also represent challenges. Furthermore, many studies identified different career expectations and assumptions between undergraduate students and managerial employees in the hospitality industry. Therefore, this study attempts to identify the perceptions of students and managerial employees' concerning an internship or CEP. Besides, in order to identify gaps, this study examines the expectations and assumptions of hospitality management students concerning their future careers, and managerial employees' perceptions of the industries needs. Finally, in the literature it was found that employers' perceptions of the hospitality management degree value are lower compared to the students' perceptions. Hence, this study attempted to identify the perceived value of hospitality management degrees amongst

hospitality management undergraduate students and the hospitality industry in New Zealand. The next chapter explains the methodology of this study.

CHAPTER 3 METHODOLOGY

3.1. Introduction

This chapter specifies the methodology and procedures utilised in this study. The statistical procedures chosen for data analysis, as well as the logic, are included. The chapter is classified into the following sections: Introduction, Participants, Research methodology (questionnaire design, survey process, analysis of data), Limitations, and Summary.

3.2. Participants

The participants of this study were students on all levels of a Bachelor of International Hospitality Management programme at AUT University and managerial employees in a large international hotel chain in New Zealand.

3.2.1 Students

In New Zealand, there are six degree programmes in the hospitality field, at two universities (AUT University and Lincoln University), three polytechnics (Wellington Institute of Technology, Otago Polytechnic, and Southern Institute of Technology), and one professional technical school (Pacific International Hotel Management School). This study focuses on students studying on a Bachelor of Hospitality Management course at universities, not at other tertiary institutions. Polytechnics and other training providers generally perform an excellent job in supporting the industry with technical skills. However, a degree qualification from a university may provide a balance of solid managerial skills and technical skills (Brien, 2004).

It was decided to approach students at two universities that offer majors in hospitality management: AUT University and Lincoln University. However, Lincoln University does not offer the same qualification as AUT University. Lincoln University offers a Bachelor of Commerce, with the ability to major in hotel and institutional management. Therefore, it was decided to remove Lincoln University from the research. Consequently, 304 students (120 1st years, 92 2nd years, and 92 3rd years) in the Bachelor of International Hospitality Management at AUT University were approached.

Table 3: Students' sample (Number of enrolled students as at March 2007)

Study year		First year	Second year	Third year	Total
Sample	Count	120	92	92	304
Sample	%	39.5%	30.3%	30.3%	100%
Total number of	Count	113	84	14	211
responses	%	53.6%	39.8%	6.6%	100%
Number of	Count	60	67	10	137
usable responses	%	43.8%	48.9%	7.3%	100%

Table 3 presents the collection data, showing the total number and percentage of usable responses. First year students' (n=113 or 53.6%) participation was higher than second (n=84 or 39.8%) and third year students (n=14 or 6.6%). The number of third year students who responded was considerably lower than the number of first and second year students, but their surveys were completed with greater accuracy. The low number of responses for third year students was a result of many third year students completing their CEP off campus, which was compulsory before the completion of their course. These students attended class only once a week. This study therefore has limited accounts of perceptions and expectations of third year students. The low number of third year respondents influenced the high level of rejected questionnaires in the survey. Furthermore, the student survey was often conducted at the end of classes. Some students may have been preoccupied with moving on to another class or leaving the university, which in turn may have resulted in the rejection of some surveys.

Out of the 211 student responses, 137 were usable. The remaining 74 responses were not used, as they were either incomplete or contained errors. For example, some students misinterpreted the instructions, which stated that only one answer was required for each question. First year students made more mistakes than did second and third year students. Many first year students ticked more than one response for a single question. The questionnaire included a complicated design, which may have contributed to the high number of unusable questionnaires.

3.2.2 Employees

The hospitality industry is a large industry and its employees represent a large workforce in New Zealand. Statistics New Zealand (2006) showed that the number of jobs in the hospitality industry (accommodation, cafes and restaurants) had increased from 80,460 (2000) to 98,560 (2005) during the five years. According to Goodchild and

Leung-Wai (2005), the summary of employee numbers in 2004 for the accommodation industry was 28,615 employees, which occupied 24.4 percent of the hospitality industry in New Zealand. Furthermore, in 2004, numbers of enterprises for the accommodation industry were 4,045 premises in New Zealand (Goodchild & Leung-Wai, 2005). Therefore, the researcher decided to select managerial employees in accommodation industry for the employees' sample in this study, as employees in the accommodation industry represent a large portion of the total number of employees in the hospitality industry. Managerial employees were selected from one large international hotel chain in New Zealand.

The initial aim was to obtain 150 employees from 10 New Zealand hotels (from one international hotel chain) as employee participants. In this study, each hotel had more than 100 rooms of a 3-4 star hotel rating operating in New Zealand. The selected international hotel chain currently operates approximately 23 hotels and resorts across key business and holiday destinations in New Zealand. On average, each hotel has fifteen managerial employees. However, the number of employees at management level was lower in some hotels, with many senior management positions, such as human resources, served by other hotels within the chain. Only seven hotels were approached, and six responded. The total number of managerial employees in the six hotels was 119. The six hotels included two from Auckland (number of respondents: n=47) one from Rotorua (n=20), one from Wellington (n=20) and two from Queenstown (n=32). These regions are the top four areas for the employment of hotel accommodation staff in New Zealand (Statistics New Zealand, 2007) (Refer to Appendix K).

Table 4: Managerial employees' sample

Regions		Auckland	Rotorua	Wellington	South Island	Total
Sample	Count	47	20	20	32	119
Sample	%	39.5%	16.8%	16.8%	26.9%	100.0%
Total	Count	35	16	8	28	87
number of responses	%	40.2%	18.4%	9.2%	32.2%	100.0%
Number of usable	Count	32	13	6	23	74
responses	%	43.2%	17.6%	8.1%	31%	100.0%

Out of 119 employees approached, 87 (73.1%) responded. This represents a high response rate, despite the fact that the survey was executed over a short period (from 15th August 2007 to 15th September 2007). Postal surveys typically receive only

twenty-five to thirty percent (Ticehurst & Veal, 2000). The reasons for a high response rate might be that six hotels are under the same International Hotel chain and the researcher sent an e-mail to each hotel human resource manager to ask for cooperation in this study. In addition, an F&B manager at one of the hotels wrote a letter of support for the research and distributed the letter to the human resource mangers involved. Response rates and usable responses of the Auckland region were higher than for the other regions. It seems that the Auckland region is a very significant destination for tourists and business people. Consequently, Auckland hotel A and B were large hotels with many employees, which generated higher numbers of usable responses. However, only 74 out of 87 responses were usable for this study. The remaining 13 managerial employee respondents made mistakes (e.g. ticking more than one box for a single question) while answering the questions.

Table 5: Frequency analysis of job positions with working department of employee

respondents

respondents										
Job position Department		Low		Mid		Тор				
levels	Super		Manag		Manag		Management		Total	
Working field	Count	%	Count	%	Count	%	Count	%	Count	%
FO/	Count	70	Count	70	Count	70	Count	70	Count	70
Reservation	7	9.5	14	18.9	5	6.8	0	0	26	35.1
F&B	7	9.5	1	1.4	5	6.8	0	0	13	17.6
House										
Keeping	6	8.1	1	1.4	7	9.5	0	0	14	.18.9
Admin &										
General	3	4.1	5	6.8	8	10.8	0	0	16	21.6
Executive										
Office	0	0	1	1.4	1	1.4	3	4.1	5	6.8
Count	23	31.1	22	29.7	26	35.1	3	4.1	74	100

The highest response rate from employees came from Mid-Managers (Department Managers), with Department Supervisor respondents ranked second. The lowest response rate came from Top Management level (GM, n=3). This was because there is only one GM Position in each hotel and only three out of six GMs responded. The highest department response rate was from the Front Office Department (35.1%). The Front Office Departments have more managerial employees (e.g. a number of assistant managers, night duty managers and one department manager) than other departments.

There were fewer respondents from low management level employees (Department Assistant Manager or Duty Manager or Graduate Management Trainee) in F&B and

Housekeeping departments than from other departments. It seems that the Front Office departments employed several Assistant Managers, but the F&B departments utilised several supervisors and one Restaurant Manager or Assistant F&B Manager in each hotel. Furthermore, some hotels did not have a Restaurant Manager; therefore, the numbers of Low Management level employees in F&B might be lower than other departments, such as Front Office, and Administration departments. The housekeeping departments also employed one department manager and several supervisors. Therefore, it was difficult to find low management level employees in the Housekeeping departments.

3.3. Research method

The survey used quantitative questionnaires. Questionnaire surveying is a survey method that stands out in contrast to qualitative techniques.

Qualitative techniques involve interaction on some level between the researcher and the respondents. They encourage the researcher to gain an in-depth appreciation of the issues and they draw detailed responses. The interaction between researcher and respondents can stimulate discussion and uncover issues that may not have been initially apparent to the researcher. However, a disadvantage of qualitative techniques is the fact the results may not be biased towards the respondents involved in the research, and may not always represent the greater population (Bryman, 2001; Veal, 1997).

In contrast, quantitative techniques offer statistically reliable results. They may determine if one idea or concept is better than the alternatives, and the results can be argued to represent the wider population. In other words, the proportion of respondents with a given response may be seen as a reflection of the proportion of the greater population that may have offered the same response if asked (Bryman, 2001; Veal, 1997). These were the key reasons for selecting a quantitative research method for this study. The researcher wanted accurate and realistic results relating to the perceptions of hospitality management students and employers. The researcher attempted to draw a wider sample size and believed a quantitative technique (questionnaire) rather than qualitative techniques would provide this opportunity. Furthermore, a questionnaire survey improves the accuracy of results, as results are relatively easy to understand and

master and the research may be easily compared to other similar studies (Eiser & van der Pligt, 1988; Veal, 1997). This method also allows the researcher to gather many data in a short amount of time using a statistical computer programme and allows for a broader study.

However, a quantitative survey also has some disadvantages. Respondents may answer inaccurately if the questionnaire takes a lot of time to complete. Hence, the questionnaire should not ask too many questions. As questionnaires are standardised, it is not possible to explain aspects of the question that may be misinterpreted by participants. A solution may be to pilot the questions on a small group of students, or with friends and (Bryman, 2001). This study included a pre-test before the data collection phase commenced, in order to ensure the questions were clear and could be interpreted as intended.

The questionnaire for students was administered directly with the support of lecturers and researchers. The questionnaire for employees in hospitality was conducted through self-completion surveys, which were mailed-out, with an enclosed self-addressed envelope. The self-completion survey allows respondents to answer whenever they are free (Bryman, 2001). For this reason, this method is considered non-intrusive compared to interviews. However, mail surveys can take several weeks to complete between mailing out questionnaires and receiving responses (Bryman, 2001; Veal, 1997).

3.3.1 Questionnaire design

Both the student and employee questionnaire surveys were created after an extensive review of the literature relating to hospitality management, travel and tourism students' career expectations, and assumptions and perceptions of employees in the hospitality industry (Johns & McKechnie, 1995; Li & Kivela, 1989; Nebel, Braunlich, & Zhang, 1994; Petrova & Mason, 2004). The questionnaire surveys were mainly adapted from a study by Petrova and Mason (2004), which was designed to identify "the value of tourism degrees, comparing students' career expectations and aspirations with industry needs and perceptions of travel and tourism degrees and graduates" (P.153). The survey was completed by 187 students from the University of Luton, and six interviews were conducted with employees of a charter airline.

The study by Petrova and Mason (2004) focused on tourism degree students. The authors were to identify the value of the tourism degrees in relation to what tourism students needed to enter the industry, in comparison to what people in the industry expected of tourism graduates. This study focused on undergraduates in hospitality management in the New Zealand hospitality industry, with a goal of identifying students' expectations and assumptions of their future working careers post graduation and the value of the hospitality management degree. In addition, the focus was to identify the background of managerial employee respondents working careers and their expectations and assumptions of a new employee with a Bachelor of Hospitality Management degree. Both the student and employee questionnaires were designed with those goals in mind.

Questionnaire for students

The students' questionnaire included 22 questions, based on the following key questions:

- Where do students expect their career to begin?
- Are students informed about the working environment of the hospitality industry?
- Are students aware of the hospitality industry needs, and of the requirements they pose to potential employees?
- What is the value of a hospitality degree to students?
- Do students believe different expectations or gaps exist between hospitality management educators and employees in the hospitality industry and that the internship or cooperative education programme will help their future career? (Refer to Appendices A & B).

Questionnaire for managerial employees

The employees' questionnaire included 20 questions based on the following key questions:

- What are the employees' working history, educational background, and career experiences?
- What do employers expect from graduate hospitality management students, during their time of employment at the hotel? What do hospitality industry employers believe to be the essential characteristics for graduates entering the

- hospitality industry?
- What are the employer's human resources needs the requirements for potential employees?
- What is the value of hospitality management degrees to employers?
- Do you recognise that there are different expectations or gaps between hospitality management educators and employees in the hospitality industry, and that an internship or a cooperative education programme will help students' careers in the hospitality industry? (Refer to Appendices C & D)

The questionnaire began with a message of appreciation and included the instructions for the survey. An information sheet was attached to the questionnaire. The information sheet introduced the survey, included a welcome message and announced the purpose of the survey (Refer to Appendix E). The introduction to the information sheet was based on Deutsch's (2006) English project (Deutsch, 2006). A welcome message and the statement of purpose for the survey were given in hope of encouraging the respondents to complete the questionnaire.

3.3.2 Surveys process

Development Pre-test

Before conducting the main data collection, pre-test surveys were completed with five students and five employees to check the structure of the questionnaire. The purpose of pre-test surveys was to test questionnaire wording, sequencing and layout. As such, a pre-test survey is important to establish whether there were any difficulties in understanding the questions posed (Veal, 1997). Through this pre-test, a number of problems were identified and adjustments were made before conducting the actual survey. The problems were largely to do with questionnaire layout and wording, and the answering choices available in the employees' questionnaire. For example, some questions needed more choices. The category 'other' was added 'to some questions' as an alternative 'to cover instances' when neither of the proposed choices was a logically possible answer.

Administration for managerial employees

Before sending the survey forms to managerial employees, e-mails were sent to Human Resources Managers in seven hotels on 12 August 2007 for approval (Refer to Appendix G). These e-mails explained the purpose of the study and why the researcher

had ask the respondent to participate. It also explained the benefits of the study and dealt with possible participant concerns. It also included a reference letter from the F&B Manager at Auckland hotel A, as verification that the study was genuine (Refer to Appendix H). E-mail responses were received from six Human Resource Managers, and questionnaires were then sent to employees. The sending of a questionnaire to employees began on the 15th of August 2007. The questionnaire survey included an information letter about the study and a self-addressed return envelope (Refer to Appendix C). Out of 119 potential employee respondents, 87 responded. The 87 responses included 35 from Auckland, 16 from Rotorua, 8 from Wellington and 28 from the South Island.

Administration for students

The students started the surveys from 6 August 2007 and were completed on 27 August 2007. Return rates for questionnaires is usually low (Bryman, 2001; Veal, 1997). However, a return rate can be dramatically improved if the questionnaire is delivered and responded to in class. Therefore, the survey was performed at the conclusion of the students' class. It took approximately 15 minutes for respondents to complete the questionnaire. The lecturer or researcher explained the purpose of the study, why the researcher chose to ask students to participate in the study, and the benefits of the study. Students were also told that there were no predictable risks connected with the project, but if they felt uncomfortable answering any questions; they could exit from the survey. Out of 304 students, 211 completed the questionnaire. First year students (n=113) participated more than second (n=84) and third year students (n=14). The first year students were more prevalent in classes, whereas many third year students were on Cooperative Education Programme (CEP) or scattered in smaller classes.

3.3.3 Analysis of data

The researcher completed analysis of the collected data once the completed student and employee questionnaires have been collated and checked. The completed student questionnaires were classified by study years and employee questionnaires were divided by workplaces. For data entry the statistical computer programme, 'Statistical Package for Social Sciences' (SPSS), Version 13 (SPSS 13) was used. SPSS is a statistics package particularly suitable to social science survey data. Therefore, it seems to be amongst the most widely used programs for statistical analysis in social sciences.

The questionnaire included multiple choices with single response and semantic rating questions. In addition, there was one open-ended question for identifying the work places of employee respondents. For those questions, the answers were categorised into like groups. For example, employees in hotels were originally from six separate hotels, and the researcher then condensed the six hotels into four regions: Auckland region; Rotorua region; Wellington region and South island region (Refer to Appendix I).

Some raw data were aggregated into significant or similar groups to reduce the vast range of answers, and to produce a more accurate data analysis (Refer to Appendix I). The transform function (recorded into different variables) of SPSS was used to merge samples of responses into a smaller number of groups in order to improve the strength of the data analysing. The data were analysed using Frequency and Cross-tabulations. A frequency distribution shows the distribution of a single variable, and a contingency table explains the distribution of at least two variables simultaneously (Bryman, 2001). On the other hand, a cross-tabulation shows the joint distribution of at least two variables. They are generally shown as a contingency table in a matrix format (Bryman, 2001). Chi-square tests were used to determine whether the variables were independent by testing the significance of the relationship between the two variables.

According to Pryman (2001),

"Chi-square is a test of statistical significance, which is typically employed to establish how confident we can be that the findings displayed in a contingency table can be generalised from a probability sample to a population" (p.501).

3.4. Limitations

The questionnaire was somewhat complicated, and perhaps as a result, some respondents made errors in their response. For example, in some instances the instructions lacked full explanations on how to answer questions. Furthermore, some questions were not clear, and respondents may have misunderstood them. However, the pilot test failed to identify these design faults, and therefore did not afford the researcher the opportunity to improve the design.

Employee participants for the survey were chosen from the hotel industry as time constraints prevented the research from exploring a broader cross section of the hospitality industry. Furthermore, the hotels that were approached were part of a very large worldwide hotel chain, as it was believed this would result in the best representation of the industry. However, it may be argued that one hotel chain cannot be considered to represent the wider hospitality industry. It is important to acknowledge, however, that each department within a hotel (such as food and beverage, front office, house keeping, and human resources) has unique characteristics that can be designed to suit different sectors of the industry, such as restaurants and bars.

The spread of job positions for employee participants was quite wide, but only three out of six GMs responded. Many hospitality students may desire to be a GM in the future, therefore the research would have benefited from the GMs' insight. Importantly, their perceptions of what they believe graduates need to accomplish in order to reach the position of GM would have added value to this study.

This study has a weak profile of hospitality managerial employees because the researcher did not investigate the respondents' gender and age, although these are critical questions in almost any survey. If the researcher had included gender and age questions, this study could have offered a stronger profile of managerial employees.

3.5. Summary

The methodology used self-completion quantitative questionnaires, created by the researcher, based on a study by Petrova and Mason (2004). The surveys were conducted by 211 students from the Bachelor of International Hospitality Management at AUT University and 119 management level employees from six international hotels in New Zealand. Before conducting the survey, pre-test surveys were conducted. The students' survey was completed over a three week period, from 6th to 27th August, while the sending of the questionnaire to employees began on 15th August, 2007. The data were analysed using the 'statistical analysis software, 'Statistical Package for Social Sciences', Version 13 (SPSS 13).

The use of a quantitative approach provides the ability to quantify determinants identified by previous research. This research provides a quantitative link between the students and employees' career expectations, assumptions and requirements. The next chapter will highlight the findings of this study.

CHAPTER 4 FINDINGS

4.1. Introduction

This chapter explores the findings from the data analysis of 137 student respondents and 74 managerial employees in the hotel industry. Questionnaires were used as the survey instrument for both students and employees. The questionnaires analysed hospitality management students' expectations and assumptions concerning their future working careers; and to investigate the working histories of managerial employees and their perception of the needs of the hospitality industry. The questionnaires were also used to consider the value of hospitality degrees for hospitality management students and their employers. The students' questionnaires included 22 questions and the employees' questionnaires included 20 questions. The findings were drawn from Frequency, Crosstabulation and Chi-square analysis of the results. Chi-square analysis allows the researcher to ascertain whether 'observed' frequencies show sufficient difference from the 'expected' frequencies in order to determine whether the two variables are, in fact, related (Wild & Seber, 2000).

The Chi-square results that were statistically significant are discussed in this report. Values of less than 0.05 were taken as statistically significant. However, values of more than 0.12 show no relationship between the variables under consideration (Bryman, 2001; Wild & Seber, 2000). As a result, if the Chi-square results indicate the level of association is not significant, the data tables show just counts and percentage.

4.2. Students' survey

This section will cover the findings of the student survey. The student data were analysed following the five main questions (Refer to Appendix B). When analysing these questions, some answering scales were merged in order to obtain more accurate results (Refer to Appendix I).

4.2.1 Students' career expectations for the future

This section presents responses to questions about students' career expectations for the future, which includes responses to the question 3, 6, 7, 16 and 17:

Q3: Do you intend to follow a career related to the hospitality industry after graduation?

Q6: After graduation, if you work in the hospitality industry, which department would be your preferred choice?

Q7: What do you think will be your entry-level first job after graduation?

Q16: What level of employment do you expect to hold five years after completing your course?

Q17: What level of employment do you expect to hold 10 years after completing your course?

Table 6: Cross-tabulation analysis of students' career expectations for the future - I

Table 0. C10.	ss tabulation analys	of students c	arcer expectations for the i	uture 1
Study year		First year	Second & Third year	Total %
Chi-square (S	Sig.) .283			/0
Expectation	after graduation			
Probably	Observed	54	73	92.7
Floodoly	Expected	56	71	92.1
Uncertain	Observed	6	4	7.3
Uncertain	Expected	4	6	1.3
Count		60	77	100.0 137

Note. Chi-square result indicates the level of association is not significant

Most student respondents wanted to work in the hospitality industry after completing their courses. Second year students were the most adamant that they would work in the hospitality industry after completing their course. Third year students indicated 'certain' less than did first and second year students. As the number of third year student respondents was very small, third year student respondents were added to the second

year student respondents to get more accurate results. Table 6 shows that the 'observed' frequencies are not significantly different from the 'expected' frequencies. Consequently, the two variables are not related; that is, there is no evidence of relationship between students' study year and expectations of a career in the hospitality industry after graduation.

Table 7: Frequency analysis of students' career expectations for the future - II

Job field/	F&B		House Keeping		Front Office		Admin. & General		Total	
department	Count	%	Count	%	Count	%	Count	%	Count	%
	51	37.2	3	2.2	33	24.1	50	36.5	137	100
Entry level job position	Front line employee		Management Trainee		Department Manager		Department Manager		Total	
Job position	51	37.2	64	46.7	14	10.2	8	5.8	137	100
5 years after graduation	Lo Manag		Mi Manage		To Manag		Self-er	nploy	Tot	al
	14	10.2	57	41.6	48	35.0	18	13.1	137	100
10 years after graduation	0	0	8	5.8	77	56.2	52	38	137	100

Table 7 shows that student respondents preferred working in Food & Beverage (F&B) departments as their first job field/department and then Administration or General, including Human Resources, Finance and Sales or Marketing Departments. Furthermore, Management Trainee level was the position students thought most likely to be their job entry level. Working at a hotel as a front line employee ranked second. Undergraduate students seemed to believe that when holding a qualification, such as a bachelor of hospitality degree, they did not need to work in the Housekeeping Department and they expected to enter a high-level position. A minority of the students expected that they would start working in a Housekeeping Department as a front line employee or a Management Trainee.

Many student respondents expected their job positions to be at the Mid Management or Top Management level within five years of graduation. A total of 57 (41.6%) student respondents expected that they would work in the hotel industry as Mid Management manager (e.g. department manager) five years after graduation. A total of 77 (56.2%) student respondents expected that they would work as a top manager in the hotel

industry whereas 52 (38%) student respondents thought that they would be selfemployed ten years after graduation. No respondents expected to work as a low management level employee in the hotel industry ten years after graduation.

The findings of students' career expectation for the future suggest that most student respondents want to work as a Management Trainee in a Food and Beverage department in the hospitality industry after completing their course. More than half (n=77 or 56.2%) of student respondents suggested that they will work as a top manager in the hotel industry, and a large number of student respondents (n=52 or 38%) expected they will be self-employed ten years after graduation.

4.2.2 Students' awareness of the working environment in the hospitality industry

This section presents responses to questions relating to students' awareness of the working environment in the hospitality industry, which includes responses to questions 9, 10 and 11:

Q9: Do you think that the hospitality industry is a growing industry with many career opportunities?

Q10: Do you think that jobs in the hospitality industry are poorly paid?

Q11: Do you think that jobs in the hospitality industry demand that you work unsociable hours?

Table 8: Frequency analysis of students' awareness of the working environment in the industry - I

Hospitality industry is a growing industry	Count	%
Agree	129	94.1
No opinion	1	0.7
Disagree	7	5.1
Total	137	100

Table 9: Cross-tabulation analysis of students' awareness of the working environment in the industry –II

Poor paymer hospitality in Chi-square (S	ndustry	Agree	No opinion	Disagree	Total %
The industry demand working unsocial hours					
Agree	Observed	82	14	9	76.60%
Agree	Expected	72	21	12	70.00 /6
No opinion	Observed	10	12	3	18.20%
No opinion	Expected	17	5	3	10.20 /0
Observed		2	1	4	5.10%
Disagree	Expected	5	1	1	3.1070
Count		94	27	16	137

Note. Chi-square result indicates the level of association is significant

Many hospitality management students were aware of the working environment in the hospitality industry. Table 8 shows that most student respondents (n=129 or 94.1%) believed that the hospitality industry is a growing industry with many career opportunities. However, table 8 shows that jobs in the industry are poorly paid (n=94 or 68.6%). A minority of student respondents (n=16 or 11.7%) disagreed that the industry is poorly paid. Furthermore, a large majority of student respondents (n=105 or 76.6%) recognised that jobs in the hospitality industry demand working unsociable hours. Of these student respondents, 82 (59.9%) students recognised that jobs in the hospitality industry demand working unsociable hours with poor payment. Many hospitality roles require shift work, so employees need to work roster shifts. As a result, employees may need to work weekends, public holidays and late at night or early in the morning. Student respondents may be aware of this working environment through their part-time working experience in the industry.

Table 9 shows that many student respondents (n=82 or 59.9%) have negative views of the working environments within the hospitality industry; for example the industry demands working unsociable hours with poor remuneration. The table also shows that

the 'observed' frequencies are significantly different from the 'expected' frequencies. Consequently, the two variables are related. Therefore, Chi-square analysis shows that there is a strong relationship between the perceptions that jobs in the hospitality industry are poorly paid and that jobs in the hospitality industry involve working unsociable hours (sig. to the .000 level). In other words, those who think the pay is poor also think the hours are unsociable. This result may highlight one of the causes of increasing employee turnover or disillusionment with work in the hospitality industry.

4.2.3 Students' awareness of what the hospitality industry needs and the requirements for employees

Question 12 and 13 are analysed together in table 10, and questions 14 and 15 are analysed together.

Q12: Which characteristic is the most important for an employee?

Q13: What do you think human resources departments consider the most important factor for a new employee?

Q14: Which of the following factors is the most important to possess as an employee in the hospitality industry?

Q15: Which skill do you think will be the most important for your career development?

Table 10: Frequency analysis of the hospitality industry needs and the requirements for

potential employees

Potentiar	<u> </u>	J												
Criteria for a new		owledg kperien		Specific Skill			Personality			Qualification		Tota	Total	
employee	Cour	nt %	. (Count	%	C	ount		%	Count	%	Count	%	
Total	55	40	.1	48	35.0		27		19.7	7	5.1	137	100	
Characteri stic for an employee	Ent	husiasr	n	Commitment		W	Willing to learn		learn	Specific Skills		Total		
Total	34	24	.8	49	35.8		47 34.3		34.3	7	5.1	137	100	
To possess as an employee		owledg kperien		Specific	Skill		Persor	nal	ity	Qualific	ation	Tota	1	
Total	43	31	.2	22	16.1		66		48.2	6	4.4	137	100	
For career develop- ment		nmuni- ation		C		uman lation			blem lving		elf- gement	Tot	al	
Total	59	43.1	19	13.9	22	16.	1 20)	14.6	17	12.4	137	100	

The findings of students' awareness of 'what the hospitality industry needs and the requirements for potential employees', suggest that they believed that knowledge and experience of the industry is the most significant skill for a new employee, and commitment is the most significant character trait for an employee. A total of 55 (40.1%) student respondents believed that Human Resource departments (HR) will consider knowledge of the industry and experience of the industry as being the most important factor for a new employee, and specific skills are ranked second by 48 (35%) student respondents. Student respondents believed that qualifications would not be considered as the most important factor for a new employee by HR. Only a minority of student respondents (5.1%) indicated that HR would consider qualifications as the most important factor for a new employee.

Student responses showed that they believed that commitment was the most important characteristic for an employee (n=49 or 35.8%), while willingness to learn was ranked as second, with 47 (34.3%) of student responses. Student respondents may think that employee organisational commitment will contribute to quality improvements and reduction of employee turnover. Student respondents suggested specific skills are not the most important characteristic for an employee. However, 48 (35%) student respondents indicated specific skills are the most important criterion for a new

employee (ranked second). This means that some departments may require specific skills. For example, finance departments may require finance skills and marketing departments may require marketing skills for a new employee. It seems students are aware that specific skills may be important for a new employee.

Furthermore, table 10 shows that personality (n=66 or 48.2%) is considered the most important factor to possess as an employee in the hospitality industry, and knowledge of the industry and experience were ranked second by 43 (31.4%) of student respondents. Communication skills (n=59 or 43.1%) are indicated as the most important skills for students' career development. Significantly, qualifications are considered a less important factor for a new employee. Hospitality roles tend to include personal interaction. Employees have to deal with many people, which can bring stress to their role. Perhaps this is why student respondents believed that personality is a very important factor, as a certain personality type may better suit interaction with the public. Furthermore, good communication skills will affect employees' career development. However, qualifications were also not ranked as such an important factor to possess as an employee in the hospitality industry.

4.2.4 Hospitality management education and existing gaps between educators and the hospitality industry

This section highlights responses to questions surrounding the nature of hospitality management education and existing gaps between educators and the hospitality industry. This relates to the student questions 4, 5, and 8.

Q4: Do you think that the hospitality management educators know the hospitality industry well?

Q5: Do you think that an internship or cooperative education programme will help your career in the hospitality industry?

Q8: Do you think that a different understanding of career expectations exists between educators and managers in the hospitality industry?

Table 11: Cross-tabulation analysis of hospitality management education and existing gaps between educators and the industry

Study year	First	year	Secon	d year	Third	l year	Total	
Educators know the industry well	Count	%	Count	%	Count	%	Count	%
Probably	57	41.6	61	44.5	10	7.3	126	93.4
Uncertain	2	1.5	5	3.6	0	0	7	5.1
No	1	0.7	1	0.7	0	0	2	1.5

Note. Chi-square result is .759 which is not statistically significant

Internship/CEP will help students	First	year	Secon	d year	Third	l year	To	tal
Agree	51	37.2	57	41.6	7	5.1	115	83.9
No opinion	8	5.8	8	5.8	1	0.7	17	12.4
Disagree	1	0.7	2	1.5	2	1.5	5	3.6

Note. Chi-square result is .078 which is not statistically significant

Different expectations exist between educators	F		G		m		75	
and the industry	First	year	Secon	d year	Third	l year	To	tal
Agree	38	27.8	40	29.2	8	5.8	86	62.8
No opinion	20	14.6	25	18.2	2	1.5	47	34.3
Disagree	2	1.5	2	1.5	0	0	4	2.9
Total	60	43.8	67	48.9	10	7.3	137	100

Note. Chi-square result is .792 which is not statistically significant

Most students believed that their educators know the hospitality industry well. All third year students indicated that their educators know the hospitality industry well and 57 (41.6%) of first year and 61 (44.5%) second year student respondents also believed this. The fact that more third year students indicate their educators know the hospitality industry well compared to first and second students can be explained as follows: third year students tend to study more specialised papers than first and second year students. Therefore, third year students will have greater evidence that their educators know the hospitality industry than first or second year students.

A large majority of student respondents (n=115 or 83.9%) believed that an internship or CEP will be helpful for their career in the hospitality industry because they might expect to have industry experience through these programmes. Only five (3.6%) students answered this question negatively. However, the data showed fewer third year students (7 out of 10 third year students, 70%) believed their careers would benefit from the CEP programme than did second (57 out of 67 second year students, 85%) or first year students (51 out of 60, 85%). This finding may be a result of the fact that 3rd year students have more experience in a CEP programme.

Student respondents believed that a different understanding of career expectations exist between educators and managers in the hospitality industry. Responses showed that a large majority of the student respondents (n=86 or 62.8%) believed a different understanding of career expectations exists between educators and managers in the hospitality industry. The fact that positive answers by third year students (8 out of 10 third year students) were significantly higher than first and second year students can be explained as follows: third year students may recognise gaps between their educators and managers as result of greater work experience in the hospitality industry. Only a few respondents (n=4 or 2.9%) believed the expectations of educators and the hospitality industry do not differ.

The overall student responses showed that their educators know the hospitality industry well and that an internship or CEP would be helpful for their career in the hospitality industry. However, student respondents suggested that a different understanding of career expectations exists between educators and managers in the hospitality industry.

4.2.5 Value of a hospitality degree to students

This section presents responses to questions concerning the value of a hospitality degree to students. The section relates specifically to questions 18, 19, 20, 21 and 22.

Q18: Do you think work experience is more important than a degree qualification for a prospective employee?

Q19: Do you think that a hospitality management degree will contribute more to the hotel business than three years experience in the hospitality industry?

Q20: Do you believe there are any benefits for a hospitality company if an employee has a hospitality management degree?

Q21: Do you think graduates with a bachelor's degree in hospitality will receive higher starting salaries than people who do not hold a degree?

Q22: Do you think that jobs in the hospitality industry need a bachelor's degree in hospitality management for promotion purposes?

Table 12: Value of a hospitality degree to students – I

Cross-tabulation analysis of experiences is more important than a hospitality degree with a hospitality degree will contribute to the hotel's business

hospitalit employee	y degree fo	mportant than a r a prospective	Agree	No opinion	Disagree	Total
A hospita	ality degree tel business	will contribute more than three years				70
Agree	Observed		38	12	20	51.1%
rigice	Expected		46	9	15	21.1 / 0
No	Observed		29	5	7	29.9%
opinion	Expected		27	5	9	<i>49.9</i> /0
Disagree	Observed		24	0	2	19.0%
Disagree	Expected		17	3	6	17.070
Count			91	17	29	137 100%

Note. The Chi-square result indicates the level of association is significant

A large majority of student respondents (n=91 or 66.4%) believed that work experience is more important than a degree qualification for a prospective employee. Only 29 (21.2%) student respondents indicated that a degree qualification is more important than experience for a prospective employee. Table 12 demonstrates that a little more than half (51.1%) of the students believed that a hospitality management degree will contribute more to the hotel business than three years of experience in the hospitality industry. This is surprising given that they were enrolled in a hospitality management degree. The Cross-tabulation analysis shows that 38 (27.7%) students indicated that experience is more important than a hospitality degree, but a hospitality management degree will contribute more to the hospitality business than experience.

The 'observed' frequencies are significantly different from the 'expected' frequencies. Consequently, these two variables are related. Therefore, the Chi-square analysis shows that there is a relationship between the statement that 'work experience is more important than a degree qualification for a prospective employee', and the statement that 'a hospitality management degree will contribute more to the hotel business than three years of experience in the hotel industry'. This means that a hospitality degree qualification is not an important factor when employers select a new employee but

employers believe that employees who have a hospitality degree qualification will contribute to their hotel business.

Table 13: Value of a hospitality degree to students- II

Cross-tabulation analysis of degree qualified employee gives benefits to employer with expectations of higher salaries

with expe	ectations of	higher salaries				
	-	ployee will receive		No		Total
higher sa			Agree	opinion	Disagree	%
Degree q	ualified em _l	ployee gives benefits to				
a compar	•					
*Chi-squa	re (Sig.)	.021				
	Observed		78	29	18	
Agree	Expected		75	29	20	91.2%
No	Observed		4	3	1	
opinion	Expected		5	2	1	5.8%
	Observed		1	0	3	
Disagree	Expected		2	1	1	2.9%
Bachelor	of hospitali	ity degree affects an				
employee	's promotio	n				
**Chi-squ	are (Sig.)	.253				
	Observed		50	15	10	-4-0/
Agree	Expected		45	18	12	54.7%
No	Observed		19	10	4	24.40/
opinion	Expected		20	8	5	24.1%
	Observed		14	7	8	
Disagree	Expected		17	7	5	21.2%
						137
Count			83	32	22	100%

Note. * Chi-square result indicates the level of association is significant

Many student respondents (n=125 or 91.2%) believed that a degree qualified employee offers benefits to a hospitality company. Furthermore, 83 (60.6%) student respondents expected that they would receive higher starting salaries than people who do not hold a degree. As a result, more than half of the students (n=78 or 56.9%) believed that degree qualified employees offer benefits to a hospitality company, and expect to have higher starting salaries than employees without a degree. The 'observed' frequencies are effectively different from the 'expected' frequencies. Consequently, these two variables are related. Therefore, the Chi-square analysis shows that there is a relationship between the statement, 'an employee with a bachelor degree in hospitality will receive higher starting salaries than people who do not hold a degree', and the statement, 'a degree-qualified employee gives benefits to a company'. Students who believed degree-

^{**}Chi-square result indicates the level of association is not significant

qualified employees offer greater benefits also believed degree-qualified employees should receive higher starting salaries.

The student responses also showed that 54.7 percent of student respondents believed jobs in the hospitality industry require a bachelor's degree in hospitality management for promotion purposes. These results show that many student respondents believed that their qualification would offer benefits to a hospitality company. However, more than half of student respondents believed that a hospitality degree would affect receiving higher starting salaries and promotion purposes. A total of 50 (36.5%) student respondents believed that a bachelor's degree in hospitality management will afford employees higher salaries and greater promotional opportunities. Conversely, a minority of student respondents (n=8 or 5.8%) indicated that a bachelor of hospitality degree will not affect an employee's promotional opportunities or generate higher salaries. These students' viewpoints suggest a hospitality degree is not important for a prospective employee.

The findings indicate there is little perceived value in a hospitality degree. Many student respondents indicate that work experience (n=91 or 66.4%) is more important than a degree qualification for a prospective employee. However, there were some positive signs in that 51.1 percent (n=70) of student respondents seemed to believe that a hospitality degree will contribute more to the hotel business than three years' experience and 125 (91.2%) of student respondents suggested a degree qualified employee offers greater benefit to a hospitality company.

4.3. Employees' survey

This section will begin with a description of employee respondents and the questions from the employees' survey, followed by the findings. The employee data was analysed following the five main questions for managerial employees (Refer to Appendices C & D). When analysing these questions, some answering scales were merged in order to obtain more accurate results (Refer to Appendix I).

4.3.1 Historical backgrounds of managerial employees

This section introduces responses to questions concerning profiles of managerial employees, which included the employee questions 4, 5 and 6.

Q3: What is your department?

Q4: How long have you been working in the hospitality industry?

Q5: In which department did you start your career in the hospitality industry?

Q6: What qualifications do you have?

Table 14: Cross-tabulation analysis of history backgrounds of managerial employees - I

Job p	osition	Department	Low	Mid	Top	Total
levels		Supervisor	Management Management		Management	%
Years	in the indust	ry				
Chi-sc	uare (Sig.) .(002				
Less	Observed	11	10	2	1	
than	00001100		10	_	-	
5	Expected	8	7	8	1	32.4%
5 to	Observed	7	11	11	0	
10	Expected	9	9	10	1	39.2%
	Observed	5	1	13	2	
11+	Expected	7	6	7	1	28.4%
		_				100%
Count	t	23	22	26	3	74

Note. The Chi-square result indicates the level of association is significant

A total of 32.4 percent (n=24) of managerial employee respondents had been working in the hotel industry less than five years. This group includes one Top Management level employee, which is a surprising finding. Achieving a Top Management level position generally requires considerable operational and managerial experience in the hospitality industry. However the aforementioned employee had been working as a top management level employee, although he had worked less than five years in the

industry. Responses show that employees who have been working from five to ten years account for 39.2 percent (n=29) of all employee respondents. However, no one in this group had worked as a management employee.

Employees who had been working more than 11 years made up 28.4 percent (n=21) of employee respondents. Of these employees, 13 have been working as Mid Management level (Department Manager) employees. Out of three Top Management employees (GMs), two had worked in the hotel industry for more than 11 years. The table 14 shows that the 'observed' frequencies are significantly different from the 'expected' frequencies consequently; these two variables are related. Therefore, the Chi-square analysis shows that there is relationship between the statement on employees' job position levels and the statement of years of work.

The employee responses showed:

- Just under half of the employees of the Department Supervisors (n=11 out of 23 or 47.8%) had been working in the hospitality industry for less than five years.
- More than half of Low Management level (n=12 out of 22 or 54.5%) had been working in the hospitality industry for five to ten years or more than 11 years.
- Half employees of Mid Management level (n=13 out of 26 or 50%) had been working in the hospitality industry for over 11 years.
- Two out of three employees (66.7%) of Top Management level had been working in the hospitality industry for more than 11 years

These show that in order to achieve higher positions, employees need long term working careers in the hospitality industry.

Table 15: Cross-tabulation analysis of history backgrounds of managerial employees

Job position levels	Depart Super		Low Management		Mid Management		Top Management		Total	
First job	Count	%	Count	%	Count	Count %		%	Count	%
Front office	3	4.1	6	8.1	5	6.8	2	2.7	16	21.7
F&B	12	16.2	8	10.8	12	16.2	1	1.4	33	44.6
Manage- ment Trainee House	1	1.4	1	1.4	1	1.4	0	0	3	4.1
Keeping	5	6.8	4	5.4	6	8.1	0	0	15	20.2
Administer General	2	2.7	3	4.1	2	2.7	0	0	7	9.4
Count	23	31.1	22	29.7	26	35.1	3	4.1	74	100

Note. Chi-square result is .893, which is not statistically significant

	Depar	tment	L	ow	Mie	d		Тор		
Qualification	Super	rvisor	Mana	gement	Manage	ement	Man	agement]	Γotal
Secondary	5	6.8	1	1.4	7	9.5	1	1.4	14	18.9
Certification/ Diploma Bachelor	10	13.5	13	17.6	10	13.5	1	1.4	34	45.9
Degree	4	5.4	6	8.1	4	5.4	1	1.4	15	20.3
Postgraduate	1	1.4	2	2.7	2	2.7	0	0	5	6.8
No qualification	3	4.1	0	0	3	4.1	0	0	6	8.1
Count	23	31.1	22	29.7	26	35.1	3	4.1	74	100

Note. Chi-square result is .627, which is not statistically significant

Nearly half the employees (n=33 or 44.6%) started their career in the F&B department. The Front Office department (n=16 or 21.7%) ranked second. Only a few employees (n=3 or 4.1%) started their working career at Management Trainee level. Also, table 15 shows an interesting finding: three Top Management managers started their careers in the Front Office department (n=2) and the F&B department (n=1). The data shows that front of house operational experience is more common for starting work in a hotel than back of house experience, such as Sales and Marketing, Finance and Human Resource Management, for those who go on to become a Top Management manager.

The responses show that employees have various qualifications. Certificate or Diploma qualifications (n=34 or 45.9%) were the predominant qualifications. The three Top Management employees, who responded to the survey, had varying levels of qualification. One held a secondary school qualification, one held a certification/diploma, and another held a degree. The table also shows some interesting findings: three Department Supervisors and three Mid Management employees had no

qualifications at all, although their positions were at managerial level. These findings suggest that qualification is not an absolute necessity for employees' career progression.

Table 16: Cross-tabulation analysis of qualification of managerial employees with current

working field/department

•	Qualification of employees		Certification/	Bachelor	Post	No	Total
employees			Diploma	Degree	Graduate	Qualification	%
Working							
field/depai	rtment						
Chi-square	.000						
Front	Observed	6	14	6	0	0	
Office	Expected	5	20	5	2	2	35.1%
F&B	Observed	3	6	3	1	0	
гав	Expected	3	6	3	1	1	17.6%
House	Observed	2	6	0	0	6	
Keeping	Expected	3	6	3	1	1	18.9%
Admin./	Observed	1	6	5	4	0	
General	Expected	3	7	3	1	1	21.6%
Executive	Observed	2	2	1	0	0	
Office	Expected	1	2	1	0	0	6.8%
Ca4							74
Count		14	34	15	5	6	100%

Note. The Chi-square result indicates the level of association is significant

Employees who work in the Administration/General department (n=16 or 21.6%) tended to have higher qualifications than employees in the other working fields/departments. Out of 21.6 percent (n=16) Administration or General Department (Human Resources, Finance, Sales and Marketing) employees, four held a Postgraduate qualification, five had a Bachelor's Degree, six had a Certificate or Diploma and one had just a Secondary School qualification. Employees in the Administration or General Department seem to require specific skills, such as finance and marketing skills. Therefore, they tend to have a higher level of education than employees in the other departments, such as the Housekeeping and F&B departments.

Six of the housekeeping employees did not have any qualifications. This finding suggests that managerial employees in housekeeping may not need tertiary educational qualifications. In addition, Executive Officers did not have higher qualifications than Low Management or Mid Management level employees did. Of the five (6.8%) of respondents who were Executive Officers, two had a secondary qualification and two held a Certificate and Diploma qualification. Only one Executive Officer had a Bachelor degree. This phenomenon might be due to current Low and Mid Management employees having more opportunities for educational advancement than current Top

Management employees (generational gap) do. The history of undergraduate degrees for hospitality management programmes in New Zealand is short. The first undergraduate degree programme, majoring in hospitality management was introduced by Lincoln University (Bachelor of Commerce: Hotel and Institution Management) in 1993 (Lincoln University, 2007). Therefore, current Top Management employees did not have the opportunities for educational advance that students do today.

The table shows that the 'observed' frequencies are sufficiently different from the 'expected' frequencies. Consequently, these two variables are related. Therefore, Chisquare analysis shows that there is a strong relationship between the qualifications of managerial employees and their working field or department. This seems to suggest that certain managerial employees may need a higher qualification (e.g. Front Office and Sales &Marketing department).

4.3.2 Expectations of managerial employees as graduate hospitality management students during their time of employment at the hotel

This section presents responses to questions concerning the expectations of managerial employees as graduate hospitality management students during their time of employment at the hotel. This related specifically to employee question 10.

Q10: What do you expect of hospitality management graduates during their employment at the hotel?

Table 17: Frequency analysis of expectation employment from graduates

	Count	Percent
Individual attributes	14	18.9
Achieves high levels of task	6	8.1
Willing to perform	16	21.6
Good general knowledge	16	21.6
Work performance effective	6	8.1
Works effectively in team work	16	21.6
Total	74	100

Table 17 shows three equal rates for employee responses: 'willing to perform' (n=16 or 21.6%), 'good general knowledge of the hotel industry' (21.6%) and 'works effectively in team work' (21.6%). Ranked second, 18.9 percent of employee respondents expected

hospitality management graduates to have individual attributes during their employment at the hotel. Six employees (8.1%) expected working units to achieve high levels of task accomplishment and another six employees expected an effective working performance from the graduates during their employment at the hotel. Therefore, these findings showed that managerial employees might expect that hospitality management graduates will be more willing to perform, have a good general knowledge, and work effectively in teamwork during their employment at the hotel.

4.3.3 The hospitality industry needs and the requirements for employees

This section presents responses to questions concerning the employer's human resource needs and the requirements for potential employees. This includes the employee answers to questions 12, 13, 14 and 15.

Q12: Which characteristic is the most important for an employee?

Q13: What do you consider the most important factor for a new employee?

Q14: Which of the following factors is the most important to possess as an employee in the hospitality industry?

Q15: Which skill do you think was the most important for your career development?

Table 18: Frequency analysis of human resource needs and the requirements for potential employees

Criteria for a new	Knowledge & Experience		Specific Skill		1	Personality		Quali	Qualification		Total	
employee	Count	%	Count	%	C	ount	%	Coun	%	Cour	nt	%
Total	14	18.9	10	13.5	5	46	62.2	4	5.4	74		100
Character- istic for an employee	Enthu	Enthusiasm Co		itment	nent Willing to		_	Specific Skills		Total		
Total	13	17.6	35	47.3	3	20	27.0	6	8.1	74		100
To possess as an employee	Knowle Exper	_	Specific Skill		1	Personality		Quali	fication		Tota	al
Total	12	16.2	13	17.6	5	48	64.9	1	1.4	74		100
For career develop- ment	Comr		Usir initiat	_		ıman ation		blem ving	Self- manager	Total		al
Total	15	20.3	21	28.4	14	18.9	12	16.2	12	16.2	74	100

Managerial employees tended to think that personality is the most important factor for a new employee. On the other hand, qualifications were not rated highly by managerial employees (n=4 or 5.4%). Furthermore, managerial employees believed that commitment was the most important characteristic for an employee (n=35 or 47.3%). Willingness to learn was ranked second, with 20 (27%) of responses. However, many employees did not consider specific skills as the most important characteristic for an employee.

A total of 64.9 percent (n=48) of the managerial employees also indicated that personality is the most important factor to possess as an employee in the hospitality industry. Only 1.4 percent of employee respondents suggested qualifications are the most important factor. Table 18 also shows that 28.4 percent (n=21) of employee respondents indicated using initiative is the most important skill for their career development.

The data suggest that employees consider commitment to be the most important characteristic for an employee, and that personality is the most important criterion for a new employee and an important factor to possess as an employee. On the other hand, qualifications were not rated as an important characteristic for an employee, or an important factor to possess as an employee in the hospitality industry.

4.3.4 Hospitality management education and existing gaps between educators and the hospitality industry

This section presents responses to questions relating to recognising the nature of hospitality management education and the gaps that exist between educators and the hospitality industry, which included the employee questions 7, 8 and 9.

Q7: Do you think that hospitality management educators know the industry well?

Q8: Do you think that an internship or a cooperative education programme (CEP) may help students' careers in the hospitality industry?

Q9: Do you think that a different understanding of career expectation exists between educators and managers in the hospitality industry?

Table 19: Hospitality management education and existing gaps between educators and the hospitality industry-I

Cross-tabulation analysis of educators know the industry well with internship/ CEP will help a student career in the industry

Internship or career in the	Agree	No opinion	Disagree	Total %	
Educators kn	ow the industry well				
Chi-square (Si	ig.) .000				
Certain	Observed	57	0	1	
Certain	Expected	54	2	2	78.4
T.T.,	Observed	11	2	0	
Uncertain	Expected	12	0	1	17.6
No	Observed	1	0	2	
NO	Expected	3	0	0	4.0
Count					74
Count		69	2	3	100%

Note. Chi-square result indicates the level of association is significant

Most managerial employees believed that the study of hospitality management is beneficial to the industry and they respected their education providers. A total of 58 (78.4%) managerial employees felt that hospitality management educators know the hospitality industry well and 69 (93.2%) of managerial employees believed that an internship or CEP may help students' careers in the hospitality industry. A total of 77 percent (n=57) of those believed that students' educators knew the industry well and an internship/CEP could assist students' career development. Only a minority of employees indicated they had no opinion or answered this question negatively.

The Chi-square analysis indicates that there is a strong relationship between the statement on what educators know and the statement relating to an internship or CEP

(sig. to the .000 level). The 'observed' frequencies are significantly different from the 'expected' frequencies. Consequently, these two variables are related. This seems to be very positive sign for recognising the nature of hospitality management education.

Table 20: Hospitality management education and existing gaps between educators and the hospitality industry-II

Frequency analysis of different career expectations between educators and managers in the hospitality industry

Different career expectations	Agree		No opinion		Disagree		Total	
	Count	%	Count	%	Count	%	Count	%
Total	51	68.9	15	20.3	8	10.8	74	100

The majority of managerial employees (n=51 or 68.9%) suggested that a different understanding of career expectation exists between educators and managers in the hospitality industry. Although, many employees answered that educators knew the hospitality industry well, they still indicated that there are existing gaps between the expectations of educators and themselves.

4.3.5 Value of a hospitality management degree to employers

This section demonstrates responses to question concerning the value of a hospitality degree to managerial employees in hotel industry. The section relates specifically to questions 11, 16, 17, 18, 19 and 20.

Q11: Do you consider a degree qualification an important criterion when you are selecting a new employee?

Q16: Do you think work experience is more important than a degree qualification for a prospective employee?

Q17: Do you think that a hospitality management degree will contribute more to the hotel business than three years of experience in the hospitality industry?

Q18: Do you believe there are any benefits for a hospitality company if an employee has a hospitality management degree?

Q19: Do you think graduates with a bachelor's degree in hospitality should receive higher starting salaries than people who do not hold a degree?

Q20: Do you think that jobs in the hospitality industry need a bachelor's degree in hospitality management for promotion purposes?

Table 21: Cross-tabulation of analysis of value of a hospitality management degree to employers - I

Degree qualification for a new employee with current job position A degree qualification as an important criteria for Certain Uncertain No **Total** a new employee **Current job position** Count Count Count **Count** % % % % levels Department Supervisor 12 16.2 5 8.1 31.1 6.8 6 23 Low Management 12 16.2 4 5.4 6 22 29.7 8.1 Mid 7 9.5 4 5.4 15 20.3 26 35.1 Management Top 3 0 0 0 0 3 4.0 4.1

13

17.6

27

36.5

74

100.0

Note. Chi-square result is .094 which is not statistically significant

34

45.9

Management

Total

Experience with degree contribution Experience is more important than degree No opinion **Total** Agree Disagree qualification **Degree will contribute** more to the hotel business Count % Count % Count % **Count %** than three years of experience Agree 14 18.9 3 4.1 6 8.1 23 31.1 9 12.2 2.7 4 15 20.3 No opinion 2 5.4 29 39.2 4 3 Disagree 5.4 4.1 **36** 48.6 **Total** 52 70.3 12.2 13 17.6 **74** 100.0

Note. Chi-square result is .340 which is not statistically significant

Less than half of the managerial employees (n=34 or 45.9%) suggested a degree qualification is an important criterion when they are selecting a new employee. On the other hand, more than half of employee respondents (54.1%) answered 'Uncertain' (17.6%) or 'No' (36.5%) for this question. There are some interesting findings: 15 out of 26 Mid Management employees believed that a degree qualification is not a significant criterion for selecting a new employee. This finding may be related to the qualifications of Mid Management employees: the findings of question 6 (Table 15) shows that only four Mid Management employees had a Bachelor's degree, and two Mid Management employees had a Postgraduate qualification. A total of 27 (36.5%) of managerial employees did not believe that a degree qualification is an important criterion for a new employee. However, Top Management employees agreed that a degree qualification is an important criterion when they are selecting a new employee.

Managerial employees (n=52 or 70.3%) considered that work experience is more important than a degree qualification for a prospective employee. Furthermore, nearly half of managerial employees (n=36 or 48.6%) believed that a hospitality management degree will not contribute more to the hotel business than having three years' experience in the hospitality industry. However, only 23 of employee respondents agreed that a degree qualification would contribute more to the hotel business than having three years' experience in the hospitality industry.

These findings indicate that a large majority of managerial employees believed that three years of experience is more valuable than a hospitality management degree for their company.

Table 22: Value of a hospitality management degree to employers – II

Cross-tabulation analysis of degree qualified employee will receive higher salaries with a degree qualified employee gives benefits to employer and a degree affects and employee's promotion

	p	romotion			
Degree qualifie higher salaries	ed employee will receive	Agree	No opinion	Disagree	Total %
Degree qualifie	ed employee gives				
benefits to a co					
*Chi-square (Si	g.) .534				
Agree	Observed	22	9	30	
Agree	Expected	21	8	31	82.4
No oninion	Observed	4	1	5	
No opinion	Expected	4	1	5	13.5
Diagona	Observed	0	0	3	
Disagree	Expected	1	0	2	4.1
Bachelor of hos	spitality degree affects an				
employee's pro	omotion				
**Chi-square (S	Sig.) .000				
A graa	Observed	10	3	2	
Agree	Expected	5	2	8	20.3
NI	Observed	5	4	3	
No opinion	Expected	4	2	6	16.2
Discourse	Observed	11	3	33	
Disagree	Expected	17	6	24	63.5
Count		26	10	38	100.0% 74

Note. *Chi-square result indicates the level of association is not significant **Chi-square result indicates the level of association is significant

Managerial employees believed that a degree-qualified employee gives benefits to their company, but salaries of a degree-qualified employee will not be different to an employee who does not hold a degree qualification. This study suggests that most managerial employees (n=61 or 82.4%) believed a qualified employee gives benefit to a hospitality company. Only three (4.1%) employees disagreed. This finding is in contrast with the findings of question 17. From question 17, managerial employees believed three years of experience would contribute more to the hotel business than a hospitality management degree. Despite this fact, managerial employees recognised that degree qualified employees can deliver benefits to a company. The responses also show that 38 (51.4%) of employee respondents disagreed that degree qualified employees should receive higher starting salaries than employees not holding a degree should. On the other hand, 26 managerial employees agreed that graduates with a bachelor's degree in hospitality should receive higher starting salaries than employees who do not hold a degree.

However, the table 22 demonstrates that 22 (29.7%) employee respondents believed a degree-qualified employee offers some benefit to a hospitality company, and degree-qualified employees should receive higher salaries than employee who does not hold a degree. The findings from these questions indicate managerial employees tended to believe that a degree-qualified employee offers benefits to company but their salaries should not be higher than an employee without a degree. Furthermore, the responses show that 47 (63.5%) employee respondents disagreed that jobs in the hospitality industry need a bachelor's degree in hospitality management for promotion purposes, although 15 (20.3%) of employee respondents agreed that a bachelor's degree in hospitality management is important for promotion purposes.

The Chi-square analysis in table 22 shows no relationship between the statement that 'a degree-qualified employee will receive higher starting salaries than people who do not hold a degree', and the statement that 'a degree qualified employee gives benefits to a company'. The 'observed' frequencies are not significantly different from the 'expected' frequencies. Consequently, these two variables are not related. On the other hand, there are strong relationships between graduates with a bachelor's degree in hospitality management being entitled to receive higher starting salaries than people who do not hold a degree and jobs in the hospitality industry requiring a bachelor's degree in hospitality management for promotion purpose (sig. to the .000 level). The 'observed' frequencies are significantly different from the 'expected' frequencies. Consequently, these two variables are related.

4.4. Summary

This chapter presented findings through an analysis of the data from students and managerial employees.

The findings of students were as follows:

- Most students (92.7%) expected to work in the hospitality industry after completing their course.
- The F&B department was ranked first as students preferred their first work role in the hotel by 37.2 percent students and 46.7 percent students expected that they would start job position as a management trainee level.
- A total of 41.6 percent of students expected that they would work in the hotel industry as mid management manager (e.g. department manager) five years after graduation and 56.2 percent students expected that they would work as a top manager in the hotel industry. However, 38 percent students thought that they would be self-employed ten years after graduation.
- Students were aware of the working environment in the hospitality industry well. For example, the hospitality industry is a growing industry (94.2%); jobs in the industry are poorly paid (68.6%) and the industry demands working unsociable hours (76.6%).
- Employee organisational commitment (35.8%) was indicated the most important character trait for an employee and knowledge and experience of the industry (40.1%) was indicated the most important factor for a new employee.
- Personality (48.2%) was indicated the most important factor to possess as an
 employee and communication skills (43.1%) were pointed to as the most
 important skills for employees' career development. However, qualifications
 were considered a less important factor for a new employee and possessing as an
 employee.
- Most students (93.4%) believed that their educators know the hospitality industry well and an internship programme or CEP would help students' career (83.9%). However, 62.8% supposed that different career expectations exist between educators and the hospitality industry.
- The value of a degree of hospitality management was identified as being of considerable importance. For example, many students (66.4%) believed that

industry experience is more important than a hospitality degree for a prospective employee. Many students (60.6%) expected that they would receive higher salaries than employees who do not hold a bachelor degree of hospitality management and almost of students (91.2%) believed that a degree-qualified employee gives benefits to a company. A total of 54.7 percent students expected that a bachelor of hospitality degree affects an employee's promotion.

The findings of managerial employees are following as:

- In order to achieve senior positions, employees need long-term working careers in the hospitality industry.
- A total of 44.6 percent of managerial employees started their career in the F&B department. Only 4.1 percent of employees started as a management trainee.
- Managerial employees have various qualifications however; Certification or Diploma qualifications (45.8%) were the predominant qualifications. 8.1 percent of employees have no qualifications. Therefore, this study identified that qualification is not an absolute necessity for employees' career progression.
- Managerial employees hope to see the following characteristics in hospitality management graduates: 'willing to perform' (21.6%), 'good general knowledge of the hotel industry' (21.6%) and 'works effectively in teamwork' (21.6%).
- Employee organisational commitment was identified the most important character trait for an employee by 47.3 percent of managerial employees.
- Personality was identified as the most important factor for a new employee (62.2%) and it was the most important factor to possess as an employee (64.9%).
- Using initiative skills (28.4%) was considered as the most important skill for employees' career development.
- Managerial employees (78.4%) believed that an internship programme or CEP is valuable to the hospitality industry and hospitality educators know the industry well. Although, a total of 68.9 percent employee respondents thought that different understanding of career expectations exist between educators and managerial employees.
- The value of a degree of hospitality management was identified as being very low. For example, 70.3 percent of employees believed that industry experience is more important than a degree in hospitality management. Furthermore, they thought three years of industry experience would contribute more to the hotel business than a hospitality degree. More than half (51.3) of employee

respondents disagreed that a degree qualified employee would receive higher salaries than an employee who does not hold a degree of hospitality management. In addition, 63.5 percent of employees thought that a bachelor of hospitality degree would not affect an employee's promotional opportunities. However, many employees (82.4%) suggested that a degree-qualified employee offers greater value to their company.

The next chapter presents the discussions of the findings of students and managerial employees in the hospitality industry. In addition, conclusions and recommendations are presented in Chapter 6.

CHAPTER 5 DISCUSSION

5.1. Introduction

This chapter will first discuss specific findings from the analysis of students and employees' answers to the questionnaires. This is followed by an account of the findings relative to the literature review. The implications of each of the major findings and issues raised will also be discussed.

5.2. Students career expectations and managerial employees working history

This section will compare and discuss responses to the student questions 3, 6, 7, 16, and 17 with employee questions 4 and 5.

5.2.1. Students' career expectations in the hospitality industry after graduation

Data from this study indicated that graduates of hospitality management degrees commonly work in the hospitality industry after finishing their course. Most student respondents (92.6%) in this study expected that they would work in the hospitality industry after completing their course. This study identified some differences and similarities to studies in the literature. Johns and McKechnie (1995) found that just over half the graduates in hospitality at the Norwich Hotel School selected careers within the hospitality industry. On the other hand, some recent studies identified that many graduates start working in the hospitality industry after completing their course (Brien, 2004; Harkison, 2004b; Jenkins, 2001). However, the Norwich study was conducted more than ten years ago so the circumstances may have changed.

Results suggest that hospitality management students want to work in the hospitality industry after completing their courses although, to them, this usually means working as a front line employee. If graduates of hospitality management courses enter the hospitality industry after finishing their studying, the industry will have well qualified employees. Graduates may be able to work more efficiently and successfully than unqualified employees may because they have a valuable insight into hospitality management from their study. As a result, hospitality organisations may be able to succeed with less training, thus reducing the organisations' financial expense. If

organisations are able to use some of that saving to fund other initiatives for improving employees' working conditions, employees may show greater motivation to work in their organisation.

5.2.2 Expectation of first job field/department and first job entry level

The results relating to students' expectations of what and where their job may be, show that of 137 student respondents, 51 (37.2%) expected that their first work department (field) would be in an F&B department in a hotel, although F&B related roles are generally physical in nature. Students might believe the F&B department is easy to enter into because they do not require as many prerequisites as do finance, human resources or a sales and marketing department. These departments require some prerequisites in prospective employees, but F&B has different requirements. Furthermore, student respondents may have experience in a F&B department as a part-time employee while studying in their course. In addition, students might recognise that the F&B department experience is one of the most important experiences as the main career route to obtaining a general manager position although this study did not examine that. However, this has been identified by many previous studies (Harper, Brown, & Irvine, 2005; Lakdin, 2000; Ladkin & Riley, 1996; Nebel, Braunlich, & Zhang, 1994; Steele, 2003). Therefore, many students seem to expect their first job field or department will be in an F&B department.

Students' expectations of first job department (field) were similar with findings of the background of employees' working career histories in this study. Of 74 managerial employees, 33 (44.6%) started in the Food and Beverage department in the hotel industry. The Front Office department was ranked second (n=16 or 21.6%). However, student respondents would prefer to work in Administration or General Departments (36.5%), more than in the Front Office departments (24.1%). One interesting finding was that only three students expected to work in the Housekeeping department after their studies. These results may demonstrate a problem in the hotel industry. If many employees prefer to work in the back of house, the front of house will have difficulties attracting staff. Therefore, the hospitality industry will need to find solutions to attract

both front of house and back of house employees. Possible solutions are presented in the recommendation section.

The results relating to students' expectations of their jobs at entry level showed Management Trainee (low level management) ranked first (n=64 or 46.7%). This result indicates that undergraduates of Hospitality Management expect that their job entry level in the hospitality industry will enable them to undertake a Management Traineeship. They might think that they have wide hospitality knowledge and skills with a degree and they feel that they should enter at low management level. This is consistent with Brien's (2004) findings. He stated that degree graduates of hospitality management in New Zealand want positions in hotels that provide a "specialised career path option with associated benefits" (para. 7). However, current positions of managerial employees in this study included 22 (29.7%) employee respondents who held positions as Assistant Managers or Management Trainees (Low Management level). Of these 22 employees, 11 had been working between five to ten years, and one had been working more than 11 years in the hospitality industry. Furthermore, only three employees started their jobs as a Management trainee. These results show that graduates of hospitality management degrees might find difficulty entering directly into a Management Trainee position (Low Management level).

Opportunities for Management Traineeship seem to be limited. In order to enter the Management Traineeship, candidates are expected to meet specific requirements, such as previous working experience in the industry and relevant qualifications. This study identified that undergraduate students had high expectations of entry-level positions, which were discordant with the reality of management pathways in the industry. The students' high expectations may have dangerous implications for students and the hospitality industry. If jobs do not meet student expectations, students may lose their motivation and may abandon or change their job. Therefore, the hospitality industry must be encouraged to communicate entry requirements to prospective students in an effort to promote realistic career expectations.

5.2.3 Expectations of long term careers

Data from this study indicated that undergraduates of hospitality management degree seem to have realistic long-term career expectations. Results show that 57 (41.6%)

student respondents expected to be in Mid Management of a hotel within five years of completing their course. Furthermore, 77 (56.2%) student respondents suggested that they would work as a Top Manager of a hotel within ten years of completing their course. The results associated with student expectations for long-term careers in this study were slightly lower than student expectations in a previous Anglo-Dutch study by Jenkins (2001). The Anglo-Dutch study found that 65 percent of students expected to be a department manager (Mid Management level) five years after graduating, and 70 percent of students expected to be employed as a general manager or corporate manager (Top Management level) ten years after completing their course (Jenkins, 2001).

However, the student results of this study relating to long-term career expectations and employee career paths are realistic. This is borne out by the working history of the managerial employees (Refer to table 15). The low management and mid management level jobs were occupied predominantly by respondents who had been working between five and ten years. Furthermore, two out of three GMs took more than 11 years to reach that position. Some earlier studies also explored the length of time taken to reach employment as a GM from when an employee was first employed (Harper, Brown, & Irvine, 2005; Jenkins, 2001; Ladkin, 2000, , 2002; Nebel, Lee, & Vidakovic, 1995; Steele, 2003). They found that reaching a GM position (Top Management) took between 9 and 15 years.

A total of 18 (13.1%) student respondents thought that they would be self-employed five years after completing their course, and 52 (38%) of student respondents answered that they would be self-employed ten years after graduating from their course. These results are considerably different from a study by Jenkins (2001). Jenkins found that no students expected to have their own business five years after completing their course, and around seven percent expected to have their own business ten years after completing their course. However, Powers and Barrows (2006) researched why students chose a degree in hospitality management. Their study found that one of the reasons for selecting a hospitality degree was the students' goal to operate their own business and to be independent in the future. The difference may be a result of cultural values and a perception in New Zealand that life as a self-employed worker is more attractive than life as a salaried employee.

5.3. Students' awareness of the working environment in the hospitality industry

This section will discuss students' responses to questions 9, 10 and 11.

Many hospitality management students were aware of the working environment in the hospitality industry. Most (n=129 or 94.2%) believed 'the hospitality industry is a growing industry with many career opportunities'. This finding supports the results of an earlier Scottish study by Barron and Maxwell (1993). They identified that postsupervised work experience students (students in the final year) had a negative perception of the industry (e.g. poor payment and demanding unsociable working hours) after having some work experience in the hospitality industry. In this study, a large majority of student respondents (n=82 or 59.8%) also knew that the hospitality industry is poorly paid and demands working unsociable hours. These negative images might affect employment in the hospitality industry. One of the New Zealand studies found that the most common reasons for employees' leaving from work in a hotel were firstly poor pay or poor salary and secondly, long working hours, unsociable work hours and shift work (McClean, 2006). Furthermore, according to Zhang and Wu (2004), "the hotel industry's social position in China has been frustrated by negative publicity and the globalisation of emerging industries such as Information Technology (IT) and banking" (p. 424). As a result, many graduates changed their jobs from the hospitality industry to other industries.

However, this study identified that undergraduates of hospitality management knew the environment of the hospitality industry well because many undergraduates have working experience in the wider hospitality industry while studying their courses. Therefore, hospitality management students have some exposure to the real lifeworking environment of the industry, which might help them to understand the hospitality industry and avoid dissatisfaction with their working career in the hospitality industry before they gain a permanent job position.

5.4. Needs and requirements of the hospitality industry – view points of students and employees

This section will compare and discuss responses to the student questions 12, 13, 14, and 15 with employee questions 12, 13, 14, 15 and 10.

5.4.1 Needs and requirement for an employee in the hospitality industry

The employees' views of the most important characteristic for an employee are very important to undergraduates of hospitality management. Students and employees had the same opinions of the most important characteristic for an employee. Commitment (students: 35.8% and employees: 47.3%) was identified as the most important factor for an employee in the hospitality industry, followed by willingness to learn (students: 34.3% and employees: 27%). A previous study stated that the hospitality industry seeks employee commitment because organisational commitment will be negatively related to employee turnover (Kazlauskaite, Buciuniene, & Turauskas, 2006). This applies not only to the hospitality industry, but also to all kinds of industries because "employees with high-level commitment are more likely to willingly contribute to the organisational performance and even do more than is expected from them" (Kazlauskaite et al, 2006, p.301). However, a high level of employee commitment seems to be quite difficult to expect, particularly from employees in the hospitality industry, because many people consider jobs in the hospitality industry to be simply work experience for other future jobs (Kazlauskaite, Buciuniene, & Turauskas, 2006).

Specific skills are not seen as such an important factor for an employee. Few respondents from both students (5.1%) and employees (9.5%) indicated that specific skills were an important factor for an employee, although many previous studies stressed that specific skills would be very important for developing a career (Baum, 2002; Christou, 1999; Kay & Russette, 2000; Raybould & Wilkins, 2005; Sigala, 2001; Stutts, 1995). Christou (1999) asserted that skills should be given priority when recruiting new staff. However, current students and managerial employees may believe specific skills will come from training, but commitment should come from employees' hearts. Therefore, hospitality companies should make an effort to enhance employee commitment and find solutions to generate employee commitment. For example,

empowerment may lead to a higher level of employee organisational commitment (Rafiq & Ahmed, 1998). If companies allow some degree of employee empowerment, employees may feel that they are valued company members. This may improve employee motivation and high level of employee organisational commitment.

Both students (n=66 or 48.2%) and employees (n=48 or 64.9%) consider personality to be the most important factor to posses as an employee in the hospitality industry. Prospective hospitality employees should consider their personalities and decide whether they are suited to the hospitality industry. If they think their personality is suitable for hospitality work and they have good positive attitudes to work well, hospitality industry knowledge and experience, the hospitality industry will be a suitable work place for them.

This study also explored managerial employee expectations of hospitality management graduates during their employment at a hotel (this question was asked only of managerial employees). The data indicated that the responses for three different answers to this question were equal. The three equal responses were:

- graduates of hospitality management work effort related to a willingness to perform
- good general knowledge of the hotel industry
- the ability to work effectively in teams during their employment at the hotel

These results may be very important information to students and education providers of hospitality management courses due to provide students' future working career in the hospitality industry.

5.4.2 Needs and requirements for a new employee in the industry

There is a gap between what managerial employees and students believe human resources departments consider the most important factor for a new employee. A total of 46 (62.2%) employee respondents in this study indicated that personality is considered the most important criterion for a new employee, followed by knowledge and experience (18.9%). Managerial employees also identified employees' personality as one very important factor for a new employee in the hospitality industry. Hospitality work may be stressful for employees because such work normally involves working

with people and working in a complex environment (Ross, 1995). Work stress can affect performance of all levels of employees but people have different feelings of intensity of stress depending in their personalities (Ross, 1995). Therefore, managerial employees may think that HR would consider personality as the most important factor for a new employee in the hospitality industry. On the other hand, 55 (40.2%) student respondents believed that knowledge of the industry and experience are the most important factors for a new employee entering the hospitality industry. Harkison (2004a) found that the hospitality industry, particularly in New Zealand, considers work experience more important than a degree in hospitality management.

This study suggests that a bachelor's qualification is not the most important factor for employment in the hospitality industry. A minority of student respondents (n=7 or 5.1%) and employee respondents (n=4 or 5.4%) indicated that qualifications are the most important criterion for a new employee. Furthermore, only a minority of student respondents (n=6 or 4.4%) and employee respondents (n=1 or 1.4%) answered that qualifications are the most important factor to possess as an employee in the hospitality industry. The reason for these findings might be related to the employee respondents' qualifications. Managerial employees might think that their qualification did not affect their career paths. Therefore, they might ignore a qualification as an important characteristic for employment and an important factor to possess as an employee in the hospitality industry. Harkison (2004a) also found that many managers in the hospitality industry believe that the hospitality degree qualification is not important for the hospitality industry in New Zealand (Harkison, 2004b).

Furthermore, students also indicated that knowledge of the industry and experience are more important than qualifications in the hospitality industry. Hospitality management students have many different personal reasons for selecting this field of study. For example, students may enjoy working with people, food, dining out, and travel, and therefore wish to study on a Hospitality Management Programme. Furthermore, students may want to operate their own business and to be independent in the future. The researcher supposes that these may be reasons for selecting study on a hospitality management programme.

5.4.3 The most important skill for an employee's career development

There were many different theories, in the literature, on the most important skill for an employee's career development (Refer to section 2.3.2). The results of the most important skill for an employee career development in this study for students and employees are also very different. Using initiative was ranked first by employee respondents while it was ranked third by student respondents. On the other hand, 59 (43.1%) of student respondents indicated that communication skills were as the most important skill for an employee career development.

The above results were similar to those of some earlier studies. Earlier studies stated that the modern manager or employee needs to be multi-skilled (Baum, 1990; Finegold, Wagner, & Mason, 2000; Jauhari, 2006). These authors stated that multi-skills can enable the management of a large number of tasks at the same time and increase the understanding of the processes involved in activities (Finegold, Wagner, & Mason, 2000; Jauhari, 2006). Furthermore, the hospitality industry has globalised rapidly over the last three decades, (Baum, Amoah, & Spivack, 1997) which has affected increasing global competition. Therefore, managers needed many skills, for example IT skills, language skills, and multi-cultural skills. As a result, employees seem to believe that using initiative was the most important skill for their career development whilst having a range of multi-skills allowed them to possess a wider field of knowledge from which to exercise their initiative.

However, communication skills were cited by many previous studies as the most important skill for career development in the hospitality industry (Jauhari, 2006; Knutson & Patton, 1992; Ladkin, 1999; Ladkin & Juwaheer, 2000; Ruddy, 1990; Stutts, 1995). Students might be influenced by their studies or literature when answering this question. Furthermore, many hospitality management students are international students who may have a heightened view of how important communication skills are, as they tend to face greater communication related challenges. In addition, jobs in the hospitality industry often require communication with many different people, so students might think that communication skills are the most significant skill for their career development.

These findings are significant for hospitality students and education providers. Education providers should refer to this result when developing hospitality study in relation to the career development needs of hospitality students.

5.5. Hospitality management education and existing gaps between educators and the hospitality industry

This section will compare and discuss students' questions 4, 5 and 8, and employees' questions 7, 8 and 9.

5.5.1 Understandings of both parties - students (or educators) and managers in the hospitality industry

A number of previous studies (Harkison, 2004b; Li & Kivela, 1989; Petrova & Mason, 2004; Ricci, 2005) found that many educators do not understand what the hospitality industry seeks from graduates of hospitality management studies, and the hospitality industry does not know what a hospitality degree consists of. These results indicate problems in the hospitality industry and education sector. The results of this study show that almost all students (n=128 or 93.4%) believe their educators know the industry well, and so do employees (n=58 or 78.4%), even if this is not as overwhelmingly as for the students. It seems to be a very positive signal for all three parties: students, educators and the hospitality industry. If hospitality educators know the industry well, they are able to teach hospitality management students in such a way that these students meet the industry's needs.

Both student (n=86 or 62.8%) and employee (n=51 or 68.9%) respondents in this study indicated that a different understanding of career progression exists between educators and managers in the hospitality industry, despite the finding that the hospitality educators are perceived as knowing the industry well. If hospitality educators and hospitality employers have different expectations, students are disadvantaged by the existing gaps between these two important parties. The educators are very important to students, as educators prepare students for entry into the hospitality industry. Therefore, in order to reduce the gaps, the educators and employers have to find solutions for developing the transition from hospitality education into the hospitality industry.

Previous studies (Harris & Zhao, 2004; Stutts, 1995) suggested some ways to reduce the gaps between hospitality employers and educators. The key to success in reducing the gaps between them appears to be a continuing discussion about curriculum, students, facilities, and faculty (Stutts, 1995). Harris and Zhao (2004) also suggest that education providers and the industry executives should consider job swapping, which would give many benefits to both sides especially in terms of understanding each other. For example, a hotel manager could work at a university as a lecturer, and a lecturer could work in a hotel as a manager in a certain department. With job swapping, managers will be provided time to address issues in the industry. In addition, if managers are working with the training providers in preparing students to enter the industry, they can offer a learning environment to help students remain competitive.

The suggestions of Harris and Zhao (2004) may offer a good solution in theory, but these will be difficult to implement. Swapping jobs will need commitment and perseverance from both parties, because it may not be easy changing their job situations. However, if they try swapping jobs even though on a part time basis, it will be a good way for them to understand each other. Furthermore, educators and hospitality employers need to reach a compromise as soon as possible, as many valuable graduates are concerned about their careers, having spent a lot of money and time studying. If they leave the hospitality industry, this industry in New Zealand will suffer, as graduates will search for better prospects overseas, as has occurred in other industries. Therefore, hospitality employers should share their changing needs, expectations, and priorities with the educators. Furthermore, educators need to balance the instruction of primary principles with the results from highly developed, marketplace research (Stutts, 1995).

5.5.2 Internship or Cooperative Education Programme

Most students of hospitality management have to complete an internship or CEP as part of the normal academic curriculum, while integrating a number of work terms into their programme. This study found that high rates of students (n=115 or 83.9%) and employees (n=69 or 93.2%) consider that an internship or CEP may help students' careers in the hospitality industry. These results mirror a Scottish study by Harper, Brown, and Irvine (2005). Harper *et al.* found that many qualified managers (75%) in Scotland indicated that a structured work experience, such as an internship or a CEP,

which is very practised and is a good way of developing personal skills. Internship or CEP is strongly supported by the industry.

Many studies identified positive and negative aspects of these programmes (Harkison, 2004a; Jauhari, 2006; Posey, Carlisle, & Smellie, 1988; Ricci, 2005; Waryszak, 1999). Jauhari (2006) and Waryszak (1999) stressed that it was difficult for students to get opportunities to develop managerial skills during the internship or CEP. Although hospitality companies seek to support students, they do not always full understand what support is needed. On the other hand, the authors (e.g. Harkison, 2004a; Posey, Carlisle, & Smellie, 1988) listed the positive aspects of internships and CEPs. For example, the programmes may be helpful when graduates gain a job in the hospitality industry. When undergraduate students have gained good work experience, they may use their work experience to evaluate their professional growth. Furthermore, the authors stressed that a CEP or an internship programme can be used as an important means of gaining experience in a practical training environment in the hospitality industry, and experience in competencies that are required in being a manager.

The majority of student respondents believed that an internship or a CEP might help their career in the hospitality industry. Third year student respondents (70%) were less positive than first year (85%) and second year (85.1%) students. The sample size of third year students was much smaller than that of students in other years, so no conclusions can be drawn from these results. However, third year students may have more work experience than first or second year students or they were doing CEP at the research time, so they may have some negative experiences of an internship or a CEP. If students are aware of these negative aspects, many of them might consider that these programmes will not help their career in the hospitality industry. If industry work environments are not matched by students' expectations, students might use their CEP or internship experience to find employment in another industry, which will disadvantage the hospitality industry.

If students are satisfied with their CEP or internship programme, this may give them the motivation to concentrate more on their study and working career. The CEP or internship experience will be a significant factor in developing hospitality students and benefits not only students, but also hospitality organisations and educators (Barron & Maxwell, 1993; Harkison, 2004a; Waryszak, 1999). Hospitality organisations may be

better prepared to retain employees and encourage their improved performance through the internship or CEP. Educators can adapt their courses to a suitable training method used by a given industry.

Waryszak (1999) recommended that both education providers and industry management internship or CEP should address the following issues:

- development of clear policies for the programmes, which might give a guide to students before a placement
- monitoring of these policies the work place which could result in more student involvement and greater cohesion between students and staff

These recommendations may be vital for making a CEP or an internship successful. If organisations and education providers offer clear structures for CEPs or internships, this will reduce the gap between students' expectations and the industry's work environment. Furthermore, reducing high work demands and time pressures are very important factors for helping match students' expectations of the industry. If students are stressed by high work demands and time pressures, they cannot demonstrate their abilities, although students want to have much work experience. Organisations and education providers should offer clear guidelines for CEPs or internships for students to succeed in their work experience programme.

5.6. Value of a Bachelor Degree of Hospitality Management Qualification

This section will compare and discuss the results from students' questions 18, 19, 20, 21 and 22, and employees' questions 11, 16, 17, 18, 19 and 20.

5.6.1 Value and benefits of a Bachelor of Hospitality Management Degree

Hospitality employers prefer industry work experience to a degree in hospitality management. This study matches the results of a number of previous studies (Harkison, 2004b) (Harkison, 2004a; Steele, 2003). Both employee (n=52 or 70.3%) and student (n=91 or 66.4%) respondents consider that work experience is more important than a Bachelor of Hospitality Management degree for a prospective employee. The students'

result was not expected, because previous studies had found that students believe that a hospitality degree can make a difference to the industry, which is why they chose to study for a degree in Hospitality Management (Brien, 2004; Harkison, 2004b). However, the results from this study are different because student respondents in this study did not believe that they needed a degree to enter the hospitality industry. Student respondents may work in hotels where senior and mid level managers have low levels of education, as is often the case in New Zealand. This exposure may lead students to believe that formal education is unnecessary as they have first hand examples of individuals who have succeed without qualifications. Therefore, student respondents might indicate that the industry working experience is more important than a Bachelor of Hospitality Management Degree for a prospective employee.

However, most student respondents (n=125 or 91.2%) believe that a hospitality company would benefit from employing employees who have a hospitality management degree. Furthermore, more than half of students (n=83 or 60.6%) also expected that they will receive higher starting salaries than people who do not hold a degree. A total of 61 (82.4%) of employee respondents also expected employees with a hospitality management degree to offer benefits to their company, although more than half of employee respondents (n=38 or 51.3%) disagreed that people with a bachelor's degree in hospitality management should receive higher starting salaries than people without this qualification. These results showed that a Bachelor of Hospitality Management degree was not considered as one of the criteria of employees' remuneration rates, which means employees could have the same rates of remuneration for staff whether they have a hospitality management degree or not.

However, 26 (35.1%) employee respondents agree that graduates with a bachelor's degree in hospitality management should receive higher starting salaries than people without this qualification. There may be a movement towards a higher percentage of managers believing that qualifications warrant higher starting salaries. Young managerial employees tend to be higher educated than older managerial employees. Therefore, it is conceivable that over time there will be higher education levels amongst managers, as the old make way for the young, and therefore a greater likelihood that managers will support higher starting salaries for educated graduates.

A total of 75 (54.7%) student respondents believed that jobs in the hospitality industry required a bachelor's degree in hospitality management for promotion purposes. However, managerial employees had a different opinion, as 47 (63.5%) of employee respondents disagreed with students' opinions. The value of a degree of bachelor hospitality management was indicated by managerial employees in this study as not an important factor for the hospitality industry. However, many previous studies stressed the importance of formal education in hospitality studies (Harper, Brown, & Irvine, 2005; Jauhari, 2006; Ladkin, 2000; Spivack, 1997). Harper et al. (2005) stated that a degree in hospitality management would help in the career development of hotel GMs. Furthermore, hospitality managers without a hospitality management qualification may have limited vision and most of their learning would occur through trial and error in the workplace (Jauhari, 2006). Their findings support a view of how important a degree in hospitality industry can be. Therefore, student respondents in this study might believe that their degree would help to get a better salary and they expected that they would have faster promotion than other employees without a Bachelor of Hospitality management degree.

5.6.2 Contribution of a degree in hospitality management in the industry

This study found that managerial employees and students have similar perceptions of the contribution of a degree in hospitality management to the hotel industry. Nearly half of employee respondents (n=36 or 48.6%) indicated that having three years' experience in the hospitality industry would contribute more to the hotel business than a hospitality management degree while 23 (31%) managerial employees indicated the opposite. On the other hand, 70 student respondents (51.1%) believed that a hospitality management degree would contribute more to the hotel industry than three years of experience in the hospitality industry although 26 (19%) disagreed with them. These results could indicate that students might have a little bit higher expectation of the value of their hospitality degree than did managerial employees.

These results show that perceptions of managerial employees in the hospitality industry do not appear to have changed regarding the abilities of graduates of hospitality management degrees, compared to findings of previous studies (Collins, 2002; Raybould & Wilkins, 2005). Collins found that from the employees' view, graduates of hospitality management held strong ideas about hospitality management, but lacked

flexibility when coping with the confused industry environment. Furthermore, students tended to have unrealistic expectations of the types of responsibilities in the hospitality industry (Raybould & Wilkins, 2005). Therefore, managerial employees in the industry tended to have a bias against hospitality students. For example, managers in the hospitality industry thought that students were over qualified but lacked experience, even for entry-level management positions. As a result, managers in the hospitality industry tended to disregard students' formal qualifications in hospitality management (Raybould & Wilkins, 2005), which is a risk to hospitality students and education providers. Educators and students need solutions to change the perceptions of managerial employees in the hospitality industry.

This section explored the value of a degree in hospitality management in New Zealand through some relevant questions. In this study, the value of a degree in hospitality management is still not recognised as an important qualification in the hospitality industry by managerial employees and even by students in the hospitality management course (Refer to section 4.2.5 and 4.3.5). These results may represent disappointment for hospitality management students may demonstrate a negative aspect of the hospitality industry. If a hospitality management degree has low value for students and the hospitality industry, prospective hospitality management students may not want to study this subject as their major. As a result, the hospitality industry may find it difficult to acquire good employees. Therefore, hospitality educators and students need solutions to change the perceptions of the industry.

5.7. Summary

This chapter discussed the similarities and differences of expectations and perceptions between undergraduates of a bachelor of hospitality management and managerial employees in the hotel industry. This study found that most students expected to work in the hospitality industry after completing their course. The F&B department was ranked first as students' first work place and a Management trainee position level was indicated as their preferred first position. The result of this study concerning students' long term career expectations was similar to the results of previous studies. Data indicated that students had realistic information for their career pathway and understand the environment of the hospitality industry. A total of 46 (33.6%) of students expected

that they would be a mid level manager five years after graduation, and a senior manager 10 years after graduation. This study also found that students and managerial employees have different perceptions of the value of a bachelor of hospitality degree.

Furthermore, there are many gaps of career expectations, assumptions and requirements existing between hospitality management students and managerial employees in hospitality industry although students and managerial employees share some common opinions. Therefore, the next chapter will present some recommendations for both students and managerial employees, and also educators and managerial employees in the hospitality industry, aiming to reduce these gaps with conclusions. Furthermore, the next chapter will demonstrate areas identified for further research.

CHAPTER 6 CONCLUSIONS AND RECOMMENDATIONS

6.1. Conclusion

The literature relating to different career expectations, assumptions and ideas on work requirements between undergraduate hospitality management students and management employees in the hospitality industry shows that many gaps exist. Furthermore, a number of studies identified that hospitality management degrees are not important in the hospitality industry and industry work experience is more important than a hospitality management degree (Harkison, 2004a; Steele, 2003). There are some potential reasons for the gaps that feature in the hospitality literature. The hospitality employers and hospitality educators in these studies appeared to have different perceptions (Collins, 2002; Harkison, 2004b; Li & Kivela, 1989; Raybould & Wilkins, 2005). Hospitality educators focus on teaching students as if they will be entering the hospitality industry at management level (Brien, 2004). However, the hospitality industry does not believe a hospitality management degree should entitle graduates to enter at this level in the industry without a certain amount of work experience (Harkison, 2004a). Furthermore, the hospitality educators and the industry managerial employees did not know exactly what they need and want from each other (Harkison, 2004a). In addition, previous studies identified that students and the hospitality employers have different perceptions of the value of hospitality management degrees (Collins, 2002; Raybould & Wilkins, 2005).

The aim of this research was to examine issues related to career expectations, assumptions and requirements of undergraduates of a bachelor hospitality management and the hospitality industry. The study sought to identify any differences in career expectations and thoughts on job requirements between undergraduates of hospitality management and managerial employees in the hospitality industry. In addition, the study was also designed to establish the perceived value of hospitality management degrees amongst hospitality management undergraduate students and managerial employees in the hospitality industry. A number of significant observations appeared from the research.

The study identified similarities and differences in perceptions between hospitality management students and managerial employees in the hotel industry. Undergraduates and managerial employees shared some common opinions. For example, they agreed that the most important factor for success as an employee was commitment and the most important factor to possess as an employee in the hospitality industry was personality. Employees with a high level of commitment are very important to hospitality organisations, as they will have a direct impact on the companies' performance. Similarly, a lack of commitment will generate higher levels of employee turnover (Kazlauskaite, Buciuniene, & Turauskas, 2006). This study shows that both students and employers recognise that commitment is the most important determinant of success for work in the hospitality industry. This shared belief bodes well for hospitality students in the future, as they can have confidence that employers will look for committed employees. Furthermore, personality is identified as a key factor for employees. The nature of the industry is such that employees are frequently involved in personal interactions, be it with customers, fellow staff, or suppliers. Employers seek the confidence to know their employees will be well perceived by customers, and personality has a large role to play in success in customer/employee interactions. Therefore, hospitality employees need to possess a positive personality. The results suggest that both students and employees know what is required for success in the hospitality industry. When employers and employees have similar beliefs and expectations, there are less likely to be disappointments that lead to turnover and dissatisfied customers.

A degree in hospitality management was not recognised as an important characteristic for a new employee and an important factor to possess as an employee in the hospitality industry by either students or employee respondents. However, both students and managerial employees recognised that aspects of hospitality education were advantageous; for example the knowledge that lecturer had of the industry was considered important and the practical was also highly regarded.

The study found a number of differences in the perception of undergraduates of hospitality management and managerial employees in the hospitality industry as far as a new employee and career development were concerned. Managerial employees argued that personality was an important criterion for a new employee while students felt that knowledge and experience in the industry were the most important. Hospitality work may be stressful for new employees because their expectations of hospitality work may differ from the reality of working life in the industry. This may lead to disappointment, dissatisfaction and ultimately turnover. Employers may witness high levels of turnover, as is common with entry level

staff in the industry, and conclude that those individuals who came and went did not have a suitable personality for hospitality. It is understandable that students rated knowledge and experience as the most important criteria for securing a job in the industry, as students are aware that work experience is valued more highly than formal education by employers. One can appreciate the pressure on graduates to gain experience in order to progress in their chosen career, as they rate experience more highly than any other criteria. The result suggests that in order to gain employment and succeed in the industry, prospective employees require work experience, combined with a positive personality.

Managerial employees argued that using initiative skills was the most important skill for an employee's career development, while students believed that communication skills are the most important. Furthermore, the study highlighted that students and managerial employees have different perceptions of the value of the hospitality management degrees. For example, students believed that a Bachelor of Hospitality Management degree would contribute more to an organisation than three years' industry experience, but managerial employees had a different opinion. Students also thought that they should receive a higher starting salary than employees who do not hold a degree. In addition, students believed that a bachelor of hospitality degree would be an important aid for promotion purposes. However, managerial employees did not agree. This result was consistent with many previous studies (Brien, 2004; Collins, 2002; Raybould & Wilkins, 2005). The gaps between the expectations of students and the beliefs of managerial employees need to be addressed by the industry. Otherwise, the likelihood is that graduates will soon become despondent as they learn that the reality of working life in the industry differs from the expectations they held as students.

The following key points emerged from the research. Most students expected that they would work in the hospitality industry after completing their course. Previous studies (Barron & Maxwell, 1993; Johns & McKechnie, 1995) identified that many students tended to change their job to other industries after completing their study. It is difficult to compare the students in this study with the results of the previous study, as the undergraduates here are yet to enter the workforce. The result of this study about students' long term of career expectations is similar to the employees' working history. Students also appeared to have a realistic understanding of the work environment in the hospitality industry. The students' perceptions of the current environment seemed to suggest a more positive situation than

that represented in the literature, which is encouraging for the hospitality industry and undergraduates of hospitality management.

This study found that even if students and managerial employees share some common opinions, many gaps still exist. Furthermore, many of the results matched the results of previous studies, so the circumstances and environment of the hospitality industry may not have changed much at all. Hence, there remain some differences of perception between hospitality management students and hospitality employers. In order to reduce the existing differences of perception between undergraduates of hospitality management and the hospitality employers, the hospitality education providers and hospitality employers may need to cooperate in addressing these different perceptions. Some recommendations to address these gaps are proposed in the next section.

6.2. Recommendations

The recommendations focus on the key areas of students' expectations and assumptions of their job entry level; the working environment in the hospitality industry; the needs and requirements of the hospitality industry; and the value of a bachelor's degree of hospitality management.

6.2.1 Students' expectations and assumptions of their job entry level

Students seem to expect their first entry-level job will be as a Management Trainee level (low management level), but the opportunities for this are very limited. Students and managerial employees believed that an internship programme and CEP might help students' careers in the hospitality industry. Therefore, one suggestion to address this problem is for hospitality organisations to accept an internship and CEP as preparation for a Management Trainee level position. Organisations could offer long term (between 6 and 12 months) internships or CEP programmes to students in order to foster operational and managerial experience and prepare students for low management level employee level positions.

Many student respondents preferred work in the back of house rather than the front of house. This may represent a challenge for the hotel industry. If highly qualified employees prefer to work in the back of house, the front of house may have difficulty with employment. Therefore, hospitality organisations may need different employee benefits to tempt students to seek employment in front of house roles as opposed to positions in the back of house. For example, hospitality organisations may need to offer employees in the front of house higher rates of remuneration and more opportunities for promotions than employees in the back of house. Prospective employees may not apply for front of house roles, as they tend to include working unsociable hours and on public holidays, with more stressful interactions with customers than employees who work back of house. Organisations may be forced to address these issues in order to attract students to front of house roles, which are essential roles in high performing hospitality organisations.

6.2.2 Working environment in the hospitality industry

Undergraduates may have difficulties adapting to working life in their first work experience, because their expectations do not match the reality of their first job role. (Waryszak, 1999). Furthermore, their working life will differ from what they learnt on their academic programmes. These differences may cause confusion, which can be stressful to students and may have a negative affect on their work career. Therefore, the hospitality industry and students would benefit from education that provides a realistic appraisal of the environment of the industry and hospitality work places. Students need the ability to cope with the obstacles associated with starting work in the hospitality industry. Waryszak (1999) suggested that aspects of the work environment need to be highlighted in preparing students for work experiences, to help match students' expectations with their company work environment. The researcher thinks this suggestion is a good idea for students and hospitality companies and may help students' first work experience.

Many students in this study seemed to understand the working environment in the hospitality industry. They expected wages and salaries in the hospitality industry to be poor, and they understood that the industry tends to require working unsociable hours. However, these factors may still cause students to lose their motivation for working in the hospitality

industry. Therefore, hospitality organisations should take action "with respect to minimum wage levels, the duration of the working week and unsociable working hours" (Baum, Amoah, & Spivack, 1997, p.223). If these issues addressed, it is conceivable that students will continue to play a vital role in the hospitality industry.

6.2.3 Needs and requirements of the hospitality industry

Students and managerial employees have different viewpoints about the demands of the hospitality industry. For example, in this study, opinions about the most important factor for a new employee in the hospitality industry differed between students and managerial employees. Students believed that knowledge and experience in the industry while managerial employees indicated personality. Students' expectations and assumptions of needs and requirements of the hospitality industry might be affected by their studying. Hospitality educators teach hospitality knowledge and skills to students so students' expectations and assumptions of needs and requirements of the industry might come from their education. However, employees' seemed to be thinking more of their companies' goals than the goals of the students. As a result, there are gaps of perception between students and managerial employees. Therefore, educators and the hospitality employers need to share information regularly through seminars or conferences or workshops in order to address the industry's changing needs and expectations. Furthermore, educators have to balance the instruction of primary principles with the results from highly developed, marketplace research (Stutts, 1995). These suggestions will assist in reducing the gap where a different understanding of career expectations exists between educators and managerial employees in the hospitality industry.

In this study, students and managerial employees have the same opinion of the most important characteristic for an employee, which is commitment. In order to enhance high levels of employee organisational commitment, organisations have to find incentives. The researcher suggests that empowerment is one good solution for enhancing employee organisation commitment. Empowerment may lead to an advanced level of employee organisation commitment which relates to offering discretion especially to front-line employees when dealing with customers (Rafiq & Ahmed, 1998). If organisations allow

some degree of employee empowerment, this may improve employee motivation, productivity and improve service for the customers (Rafiq & Ahmed, 1998). Therefore, Rafiq and Ahmed state that service by employees who have been empowered will meet customers' wants, and both employees and customers will be satisfied, which may affect employees' organisational commitment and improve company profits.

6.2.4 Value of a Bachelor Degree of Hospitality Management qualification

Managerial employees seem to place little value on bachelors' degrees of hospitality management (Collins, 2002; Harkison, 2004a; Raybould & Wilkins, 2005). Managerial employees in hotels prefer hotel industry experience than a hospitality management degree for prospective employees. Currently, many undergraduates of hospitality management work in the hospitality industry as part time employees while studying but they cannot afford to work for long hours because of their study obligations. Therefore, students' work experiences cannot satisfy the employers when they apply for positions. Furthermore, the Bachelor of Hospitality study curriculum tends to focus more on managerial levels than operational levels. As a result, students' expectations of their careers are higher than those of hospitality employers.

However, the literature identifies that operational levels experience is very important if an employee wishes to enter senior management levels. Students and educators need to know the expectations and requirements of the hospitality industry employing graduates as new employees. Furthermore, educators need to provide course content that matches the industry's expectations and requirements. For example, educators should negotiate with hospitality employers to offer long-term internships or CEP in order to support students' operational and managerial experience. Improvements in the communication between industry professionals and hospitality educators may add value to internship or CEP programs and allow hospitality organisations to identify students with real ability and proven experience. Furthermore, job swapping between educators and managerial employees may help to bridge the gaps that exist between the two parties and foster cooperation for the benefit of the industry (Harris & Zhao, 2004). These solutions will

increase the value of a bachelor degree of hospitality management for both students and those in the hospitality industry.

6.3. Areas identified for further research

The student participants were limited to students in one programme, the Bachelor of International Hospitality Management at AUT University. Further study needs to involve more students from Bachelor of Hospitality Management courses from other universities and institutions, in order to produce data that are more accurate.

The proportions of student participants were not even (Refer to Table 2). There were 60 first year students, 67 second year students, and only 10 third year students. Therefore, it will not be possible to compare thoroughly perceptions and expectations of the different levels, although many previous studies have identified that each year's students had different perceptions. Further study should select the same or similar numbers of student sample from each year to identify how they have different perceptions and expectations of their future careers.

The employee participants for the survey were chosen from only one hotel chain, although the hospitality industry includes more than just hotels. As a result, the analysis may be limited and might offer biased results. Therefore, further study should expand this area to incorporate a more advanced critical analysis.

The job positions of employee participants were quite wide ranging, but only three GMs participated. Many hospitality students may want to be a GM in a hotel in the future. Therefore, it is important for research to identify their background and working experiences. However, it was difficult to identify GMs' working careers, their perceptions and what they want to graduates when they employ graduates, as so few participated. Further research should focus more on GM positions, because students are very interested in that position.

This study focused on career expectations and assumptions, and different perceptions of undergraduates and managerial employees in hotel industry. Further study should focus on the perceptions of education providers and managerial employees in the hospitality industry,

which may help them, gain an understanding of each other. The results will be valuable to both educators and students, and the industry itself.

Many student respondents indicated that knowledge of the industry and experience are more important than qualifications in the hospitality industry. Therefore, the researcher wonders why they are studying hospitality management, if they believe the qualification is not an important factor for a new employee. This is an interesting question and so future studies need to examine why students choose this particular course of study.

Many of the questionnaires were found to be 'unusable' in the survey of both students and employees: 35.1 percent of student respondents and 14.9 percent of employee respondents did not answer completely or made mistakes while answering a question. The student survey was conducted at the end of some hospitality classes, so some students may have been eager to move on to other classes, places or activities, rather than to complete questionnaires. Furthermore, some student and employee participants seemed to have misread or misunderstood the instructions, which may account for the mistakes observed in responses. Further research should use a more carefully designed questionnaire, offer clear instructions, and include simple and logical questions.

This study identified limitations in the research and offered suggestions for future research. This study has hopefully contributed to reducing the gaps of perception between the students and hospitality employers, and also between education providers and hospitality employers. Furthermore, it is hoped that this study has promoted a greater understanding of the expectations and assumptions of those who work in the hospitality industry and positions who aspire to gain there.

REFERENCES

Reference List

- Accorhotels. (2007). Graduate management programme: Retrieved March 15, 2007 from World Wide Web: http://www.accorhotels.co.nz/careers/management.aspx.
- Barron, P., & Maxwell, G. (1993). Hospitality management students' image of the hospitality industry. *International Journal of Hospitality Management*, 5(5), v-viii.
- Barrows, C. W., & Bosselman, R. H. (1999). *Hospitality management education*. America: Haworth Press.
- Baum, T. (1990). Competencies for hotel management: Industry expectations of education. *International Journal of Contemporary Hospitality Management*, 2(4), 13-16.
- Baum, T. (2002). Skills and training for the hospitality sector: A review of issues. *Journal of Vocational Education and Training*, 54(3), 343-363.
- Baum, T., Amoah, V., & Spivack, S. (1997). Policy dimensions of human resource management in the tourism and hospitality industries. *International Journal of Contemporary Hospitality Management*, 9(5/6), 221-229.
- Brien, A. (2004). Value of a degree education: Retrieved June 5, 2007 from the World Wide Web: http://www.lincoln.ac.nz/story4477.html.
- Brownell, J. (1994). Women in hospitality management: General managers' perceptions of factors related to career development. *International Journal of Hospitality Management*, 13(2), 101-117.
- Bryman, A. (2001). Social Research Methods. Oxford: Oxford University Press.
- Buergemeister, J. (1983). Assessment of the educational skills and competencies needed by beginning hospitality managers. *Hospitality Education and Research Journal*, 8(1), 38-53.
- Christou, E. (1999). Hospitality management education in Greece: Overview and qualitative assessment. *Tourism Management*, 20 (6), 683-691.
- Collins, A. B. (2002). Are we teaching what we should? Dilemmas and problems in tourism and hospitality management education. *Tourism Analysis*, 7(2), 151 163.
- Connor, H., & Pollard, E. (1996). What do graduates really do? (No. 308). Brighton: Institute of Employment Studies.
- Damitio, J. W. (1988). Importance of managerial accounting skills to lodging managers. *Hospitality Education and Research Journal*, 12(2), 287-292.
- Damitio, J. W., & Schmidgall, R. (1991). A comparison of hospitality executives', educators', and students' views on the importance of accounting skills. *International Journal of Hospitality Management*, 10(3), 219-228.
- Damitio, J. W., & Schmidgall, R. (1993). Hospitality professional responses to ethical situations. *The Cornell Hotel and Restaurant Administrative Quarterly*, 34(4), 40-43.
- Deutsch, N. (2006). Student questionnaire on reading comprehension tests for ESL/EFL and Anxiety: Retrieved November 19, 2006 from World Wide Web: http://www.nelliemuller.com/Student_Questionnaire_and_Reading_Anxiety_in_ES L.html.
- Eiser, R., & van der Pligt, J. (1988). Attitudes and decisions. London: Routledtge.
- Finegold, D., Wagner, K., & Mason, G. (2000). National skill-creation systems and career paths for service workers: hotels in the United States, Germany and the United Kingdom. *International Journal of Human Resource Management*, 11(3), 497-516.
- Goodchild, M., & Leung-Wai, J. (2005). *Detailed profile of the hospitality industry* (Report). Wellington, New Zealand: Business and Economic Research Limited.

- Harkison, T. (2004a). *Hospitality Degrees in New Zealand: Exploratory Research*. Published Masters Thesis the Master of Education (Adult Education), Massey University, Wellington
- Harkison, T. (2004b, 8-10 December). Who pushed who? The role of an academic degree in the hospitality industry within New Zealand in Smith, K.A. and Schott, C. (2004) eds. Paper presented at the New Zealand Tourism and Hospitality Research Conference Wellington.
- Harper, S., Brown, C., & Irvine, W. (2005). Qualifications: a fast-track to hotel general manager? *International Journal of Contemporary Hospitality Management*, 17(1), 51-64.
- Harris, K. F., & Zhao, F. (2004). Industry internships: feedback from participating faculty and industry executives. *International Journal of Contemporary Hospitality Management*, 16(7), 429-435.
- Hsu, J. F., & Gregory, S. (1995). Developing future hotel managers in Taiwan: From an industry viewpoint. *International Journal of Hospitality Management*, 14(3/4), 261-269.
- Jameson, S., & Holden, R. (2000). 'Graduateness' Who Cares? Graduate identity in small hospitality firms. . *Education* + *Training* 42 (4/5), pp. 264-271.
- Jauhari, V. (2006). Competencies for a career in the hospitality industry: an Indian perspective. *International Journal of Contemporary Hospitality Management*, 18 (2), 123-134.
- Jayawardena, C. (2001). Challenges in international hospitality management education. *International Journal of Contemporary Hospitality Management*, 13(6), 310 315.
- Jenkins, A. K. (2001). Making a career of it? Hospitality students' future perspectives: an Anglo-Dutch study. *International Journal of Contemporary Hospitality Management. Bradford*, 13(1), 13.
- Johns, N., & McKechnie, M. (1995). Career demands and learning perceptions of hotel and catering graduates ten years on. *International Journal of Contemporary Hospitality Management*, 7 (5), 1-12.
- Kay, C., & Russette, J. (2000). Hospitality-management competencies. *Cornell Hotel and Restaurant Administration Quarterly*, 41, 52-63.
- Kazlauskaite, R., Buciuniene, I., & Turauskas, L. (2006). Building employee commitment in the hospitality industry. *Baltic Journal of Management*, 1(3), 300-314.
- Knutson, B. J., & Patton, M. E. (1992). How prepared am I to succeed in the hospitality industry? What the students are telling us. *Hospitality and Tourism Education*, 4(3), 38-43.
- Ladkin, A. (1999). Hotel general managers: A review of prominent research themes. *International Journal of Tourism Research*, 1(3), 167-193.
- Ladkin, A. (2000). Vocational education and food and beverage experience: issues for career development. *International Journal of Contemporary Hospitality Management*, 12(4), 226-233.
- Ladkin, A. (2002). Career analysis: a case study of hotel general managers in Australia. *Tourism Management*, 23(4), 379-388.
- Ladkin, A., & Juwaheer, T. (2000). The career paths of hotel general managers in Mauritius. *International Journal of Contemporary Hospitality Management*, 12(2), 119-125.
- Ladkin, A., & Riley, M. (1996). Mobility and structure in the career paths of UK hotel managers: a labour market hybrid of the bureaucratic model? *Tourism Management*, 17(6), 443-452.

- Lee, S. (2006). A comparison of student perceptions of learning in their co-op and internship experiences and the classroom environment: A study of hospitality management students. Published Doctor of Education in the Department of Educational Research, University of Central Florida Orland, Florida.
- Ley, D. A. (1980). The effective GM: Leader or entrepreneur? *The Cornell Hotel and Restaurant Administrative Quarterly*, 21(3), 66-67.
- Li, L., & Kivela, J. (1989). Different perceptions between hotel managers and students regarding levels of competency demonstrated by hospitality degree graduates. *Australian Journal of Hospitality Management*, 5 (2), 47-54.
- Li, L., & Leung, R. W. (2001). Female managers in Asian Hotels: Profile and career challenges. *International Journal of Contemporary Hospitality Management*, 13(4), 189-196.
- Mason, G. (1995). The graduate supply shock: Recruitment and utilisation of graduates in British Industry. London: National Institute for Economic and Social Research.
- McClean, C. (2006). An examination of managers' perceptions of turnover and retention in the hotel industry: Are hotel manager doing enough to address the issue of turnover? Published Masters Dissertation the Master of International Hospitality Management, AUT University, Auckland.
- Ministry of Tourism. (2008a). Forecasts Commentary Retrieved December 27, 2008 from the World Wide Web: http://www.tourismresearch.govt.nz/Data--Analysis/Forecasts/2008---2014-Forecasts-National/Commentary/.
- Ministry of tourism. (2008b). Forecasts Expenditure Summary Retrieved December 27, 2008 from the World Wide Web: http://www.tourismresearch.govt.nz/Data--Analysis/Forecasts/2008---2014-Forecasts---National-Expenditure/Forecasts--Expenditure-Summary/.
- Nebel, E., Braunlich, C. G., & Zhang, Y. (1994). Career paths in American luxury hotels: Hotel food and beverage. *International Journal of Contemporary Hospitality Management*, 6(6), 3.
- Nebel, E., Lee, J., & Vidakovic, B. (1995). Hotel general manager career paths in the United States. *International Journal of Hospitality Management*, 14(3/4), 245-260.
- New Zealand Education guide. (2007). Study International Tourism and Hospitality: Retrieved September 7, 2007 from World Wide Web: http://www.newzealandeducationguide.com/study/hospitality_and_tourism.htm.
- O'Mahony, G. B., & Sillitoe, J. F. (2001). Identifying the perceived barriers to participation in tertiary education among hospitality employees. *International Journal of Contemporary Hospitality Management*, 13(1), 21 29.
- Petrova, P., & Mason, P. (2004). The value of tourism degrees: a Luton-based case study. *Education + Training*, 46 (3), 153-161.
- Posey, L. O., Carlisle, K. E., & Smellie, D. C. (1988). An internship case study: How internships can benefit the student, the university, and the organisation. *Training and Development Journal*, 42(2), 59-64.
- Powers, T., & Barrows, C. W. (2006). *Introduction to the hospitality industry* (6th ed.). New Jersey: John Wiley & Sons, Inc.
- Rafiq, M., & Ahmed, P. K. (1998). A customer-oriented framework for empowering service employees. *The Journal of Service Marketing*, 12(5), 379-396.
- Raybould, M., & Wilkins, H. (2005). Over qualified and under experienced: Turning graduates into hospitality managers. *International Journal of Contemporary Hospitality Management* 17(3), 203-216.

- Ricci, P. (2005). A comparative analysis of job competency expectations for new hires: The relative value of a hospitality management degree. Published a dissertation submitted in partial fulfillment of the requirements for the degree Doctor of Education University of Central Florida Orland, Florida
- Riley, M. (2005). Viewpoint food and beverage management: A review of change. *International Journal of Contemporary Hospitality Management*, 17(1), 88-93.
- Riley, M., & Turam, K. (1989). The career paths of UK hotel managers: A developmental approach. *Signet Quarterly*, 1(1), 1-13.
- Ross, G. F. (1995). Work stress and personality measures among hospitality industry employees. *International Journal of Contemporary Hospitality Management*, 7(6), 9-13.
- Ruddy, J. (1990). Patterns of hotel management development in South East Asia. *Hospitality Research Journal*, 14(2), 349-361.
- Sigala, M. (2001, 24-27 May). *Measuring students' variables for developing e-learning platforms*. Paper presented at the 1st National Conference on Open and Distance Education, Patras, Greece.
- Spivack, S. (1997). A consensus model approach for assessing gaps between education system output and human resource demands in the tourism and hospitality sector to aid in the attainment of quality service goals. Published DPhil thesis, University of Buckingham, Buckingham.
- Statistics New Zealand. (2007). Accommodation Pivot Tables: Retrieved December 8, 2007 from World Wide Web: http://www.stats.govt.nz/tables/accommodation-pivot-tables.htm.
- Steele, R. (2003). Career analysis: a study of low, mid, and high level managers in the hotel industry in Auckland, New Zealand. Paper presented at the Sociological Association of Aotearoa New Zealand Annual Conference, 9-11 December 2003, Auckland University of Technology, Auckland.
- Stutts, A. (1995). Viewpoint: Higher education in hospitality management. *International Journal of Contemporary Hospitality Management*, 7(6), ii, 2.
- Tang, A. (2006). A warm welcome for aspiring hosts Langham Hotel is offering Internships to 20 students who want to join the hospitality industry. *South China Morning Post*, p. 5.
- Ticehurst, G. W., & Veal, A. J. (2000). Business research methods: A managerial approach. Australia: Longman.
- University of Central Florida. (2007). Experiential Learning University of Central Florida: about cooperative education: Retrieved May 9, 2007, from World Wide Web: http://www.coop.ucf.edu/?go=aboutcoop.
- Veal, A. J. (1997). *Research methods for leisure and tourism: A practical guide* (Second ed.). Edinburgh Gate: Pearson Education Ltd.
- Wagner, G. (1993). Building marketing muscle. *Lodging Hospitality*, 46(3), 34-36.
- Waryszak, R. Z. (1999). Students' expectations from their cooperative education placements in the hospitality industry: an international perspective. *Education* + *Training*, 41(1), 33-40.
- Wild, C. J., & Seber, G. (2000). *Chance encounters: a first course in data analysis and inference*. Danvers: John Wiley & Sons, Inc.
- Wood, J., Zeffane, R., Schermerhorn, J. R., Hunt, J. G., & Osborn, R. N. (2001). Organisational behaviour - A global perspective. Australia: John Wiley & Son Australia Ltd.

APPENDICES

Appendix A – Survey Questionnaire for students

Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences



Survey Questionnaire for Students at Auckland University of Technology

Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences

Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences



Instruction

Thank you again for the time you are going to spend filling in this questionnaire. Please tick your answer in the appropriate box. Only one answer is needed for each question. If you have any questions about any of the questions, please contact Ginny Kim at 021-3868 55 or email junkimb3@aut.ac.nz. Once you have completed this questionnaire please return it in the self addressed envelope that is attached to this questionnaire.

The questionnaire will include the following;

- 1) What level of management do students expect their career to begin?
- 2) Are students aware of the work environment in the hospitality industry?
- 3) Are students aware of what the hospitality industry needs are and they pose the potential employees?
- 4) What does the value hospitality degrees have to students?

Please tick the appropriate box. Only one answer is possible for each question.

1. Which school do you study at?

1) Auckland University of Technology	
2) Lincoln University	

2. Which year of hospitality management study are you currently in?

1) First year	
2) Second year	
3) Third year	

Definitely	Probably	Uncertain	Unlikely	No
4 Do vo	a think that the k	osnitality or hotel	management ed	ucators know about t
•	llity industry well		management ed	dedicis know about t
Definitely	Probably	Uncertain	Unlikely	No
	-		,	,
5. Do yo	u think that inter	nship or cooperativ	ve education pro	ogramme will help yo
career	in the hospitality	industry?		
Strongly Agre	e Agree	No Opinion	No Opinion Disagree St	
		'	1	-
6. After g	graduation, if you	work in the hospi	tality industry, v	which department wou
be you	r preferable choic	e?		
Fo	ood and Beverage			
Н	ouse keeping			
17.				
Fı				
	ales and Marketin	g		
Sa	nles and Marketin nance	g		
Sa		g		

Line	e level employee			
Mai	nagement trainee			
Dep	artment superviso	r		
Dep	artment manager			
Oth	er (Please write)			
8. Do you	think that a diffe	rent understandin	g of career exp	pectations exist betwe
educator	s and manager in t	he hospitality ind	ustry?	
Strongly Agree	Agree	No Opinion	Disagree	Strongly Disagree
9. Do you opportun	•	pitality industry i	s a growing in	dustry with many care
Strongly Agree	Agree	No Opinion	Disagree	Strongly Disagree
	hink that jobs in th	ne hospitality indu	stry are poorly	paid?
Strongly Agree	Agree	No Opinion	Disagree	Strongly Disagree
11. Do you	think that jobs in	the hospitality in	dustry demand	you working unsocial
hours?				
Strongly Agree	Agree	No Opinion	Disagree	Strongly Disagree
		1		

7. What do you think will be your entry level at your first job after graduation?

12. Wh	ich characteristic is the most important for an em	ployee?	
	Enthusiasm		
	Dedication		
	Commitment		
	Willing to learn		
	Hardworking		
	Determination		
	Specific Skills (customer service skills,		
	communication skills, IT skills etc)		
	Other (Please write)		
			'
13. Wh	at do you think human resource departments con	nsider the mo	est important factor
for	a new employee?		
	Knowledge of the industry		
	Personality		
	Specific Skills (customer service skills,		
	communication skills, IT skills and etc)		
	Qualifications		

Experience

Other (Please write)

14	. Which	of the	following	factors	is the	most	important	to	possess	as ar	emplo	yee in
	the hos	pitality	y industry?	•								

Knowledge of the industry							
Specific skills (customer service skills,							
communication skills, IT skills etc)							
Personality							
Qualifications							
Experience							
Attitude							
Commitment							
Other (Please write)							

15. Which skill do you think will be the most important for your career development?

Communication skills	
Using initiative skills	
Human relations skills	
Food and beverage skills	
Problem solving skills	
Self-management skills	
To be multi-lingual skills	
Front office skills	
Other (Please write)	

3	ırse?						
De	partment superviso	or level					
Dej	partment manager	level					
Ser	Senior management level						
Toj	o management leve	el					
Sel	f-employed						
Oth	ner (Please write)		I				
17. What le	vel of employmer	nt do you expect t	o hold after 10	0 years after comp			
your cou	ırse?						
Dej	partment manager	level					
Ser	nior management le	evel					
Тој	o management leve	el					
Sel	f-employed						
Oth	ner (Please write)						
18. Do you	think work experi	ience is more imp	ortant than a d	egree qualification			
·	think work experi	ience is more imp	ortant than a d	egree qualification			
·	tive employee?	No Opinion	ortant than a d				
prospect	tive employee?						
prospect	tive employee?						
prospect Strongly Agree	Agree	No Opinion	Disagree	Strongly Disag			
prospect Strongly Agree 19. Do you	Agree think that a hosp	No Opinion	Disagree ent degree will	Strongly Disagranger contribute to the			
prospect Strongly Agree 19. Do you	Agree think that a hosp a more than three y	No Opinion pitality manageme	Disagree ent degree will	Strongly Disag			

16. What level of employment do you expect to hold after five years after completing

20. Do	you	believe	there	are	any	benefits	for	an	employee	having	a	hospitality
mar	management degree for hospitality organisations?											

Strongly Agree Agree		No Opinion	Disagree	Strongly Disagree	

21. Do you think graduates with a bachelor's degree in hospitality will receive higher starting salaries than people who do not hold a degree?

Strongly Agree Agree		No Opinion	Disagree	Strongly Disagree	

22. Do you think that jobs in the hospitality industry need a bachelor's degree in hospitality management for promotion purposes?

Strongly Agree	Agree	No Opinion	Disagree	Strongly Disagree

Thank you very much for taking the time to complete this questionnaire.

This survey will be valuable for developing students' abilities and greatly improving the industry standards in New Zealand.

Appendix B – Primary Research Questions and Survey Questionnaire for Students

Primary Research Questions	Survey Questionnaire
1) Where do students expect their	3. Do you intend to follow a career related to
career to begin?	the hospitality industry after graduation?
	6. After graduation, if you work in the
	hospitality industry, which department would
	be your first choice?
	7. What do you think will be your entry level
	for your first job after graduation?
2) Are students aware of the working environment in the hospitality industry?	9. Do you think that the hospitality industry is a growing industry with many career opportunities? 10. Do you think that jobs in the hospitality industry are poorly paid? 11. Do you think that jobs in the hospitality industry demand working unsociable hours?
3) Are students aware of what the	12. Which characteristic is the most important
hospitality industry needs are and of	for an employee?
the requirements, they pose to	13. What do you think human resource
potential employees?	departments consider the most important factor
	for a new employee?
	14. Which of the following factors is the most
	important to possess as an employee in the
	hospitality industry?
	15. Which skill do you think will be the most
	important for your career development?
4) What is the value of a hospitality	16. What level of employment do you expect to

degree to students?	hold 5 years after completing your course?
	17. What level of employment do you expect to
	hold 10 years after completing your course?
	18. Do you think work experience is more
	important than a degree qualification for a
	prospective employee?
	19. Do you think that a hospitality management
	degree will contribute more to the hotel
	business than three years' experience in the
	hospitality industry?
	20. Do you believe there are any benefits to a
	hospitality company if
	an employee has a hospitality management
	degree?
	21 Do you think graduates with a bachelor's
	degree in hospitality will receive higher starting
	salaries than people who do not hold a degree?
	22. Do you think that jobs in the hospitality
	industry need a bachelor's degree in hospitality
	management for promotion purposes?
5) Do students believe that some	4. Do you think that hospitality management
different expectations or gaps exist	educators know the hospitality industry well?
between hospitality management	
educators and employees in the	8. Do you think that a different understanding
hospitality industry? Do students	of career expectations exists between educators
believe that the cooperative	and managers in the hospitality industry?
education programme will help their	
working careers in the future?	
	5. Do you think that an internship or
	cooperative education programme will help
	your career in the hospitality industry?

Appendix C – Survey Questionnaire for employees



Survey Questionnaire for the Hotel Industry Employees

Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences

Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences



Instruction

Thank you again for the time you are going to spend filling in this questionnaire. Please tick your answer in the appropriate box. Only one answer is needed for each question. If you have any questions about any of the questions, please contact Ginny Kim at 021-3868 55 or email junkimb3@aut.ac.nz. Once you have completed the questionnaire please return it in the attached self addressed envelope.

This questionnaire will include the following:

- 1) Employees' work history, educational background, and experiences in their career.
- 2) What industry expects of graduates in hospitality management during their employment at the hotel?
- 3) What are the human resource needs of the employers and the requirements they pose to potential employees?
- 4) Where do employers see the point of entry of hospitality and hotel management graduates in the industry?
- 5) What is the value of a hospitality or hotel management degree to employers?

It will not be used to identify you as an individual

1.	What hotel do you work in? (Please write)				
Please tio	ck the appropriate box. Only one answer is possible f	or each question.			
2.	What best describes your job? Department Supervisor				
	Assistant Manager				
	Department Manager				
	General Manager				
3.	What is your department?				
	Reservations				
	Sales & Marketing				
	Banquet Operation or Conferences				
	Food & Beverage				
	Kitchens				
	Rooms and Front Desk				
	Guest Services (Door person, Concierge, Telephones)				
	Housekeeping				
	Purchasing and Stores				
	Engineering				
	Human Resources				
	Executive Office				
	Finance and Accounts				
	Other (Please write)				

4	How lon	σ have v	ou been	working i	in the	hospitalit	y industry?)
╅.	TIOW IOII	g nave y	ou occii	WUIKIIIg I	in the	nospitant	y maasa y :	

Less than 1 year	
1 -2 years	
2–4 years	
5-7 years	
8-10 years	
11-15 years	
16-20 years	
21-25 years	
26 – 30 years	
More than 30 years	

5. Which department did you start your career in the hospitality industry?

Food and Beverage	
Housekeeping	
Rooms and Front Office	
Sales and Marketing	
Finance, Accounts, and Credit	
Human Resources	
Kitchens	
Engineering	
Guest Services (Door Person, Concierge,	
Telephones)	
Purchasing and Stores	
Other (Please write)	

	Seco	ndary School Ce	ertificate		
7	Voca	tional Qualifica	tion (Trade, New		
2	Zeala				
]	Poly	technic Certifica	ite		
]	Poly	technic Diploma	1		
1	Univ	ersity Certificate	e		
1	Univ	ersity Diploma			
]	Postg	graduate Qualific	cation		
(Othe	r (Please write)			
7. D	o yo	u think that hosp	pitality managemen	t educators kno	ow the industry well?
Finitaly,		Duohohly	Uncertain	Unlikely	No
finitely		Probably	Officertain	Cillikely	NO
		Probably	Oncertain	Omikery	NO
8. D	•	ou think that an i		erative educati	
8. D	tuder	ou think that an i	nternship or a coop	erative educati	on programme may h Strongly Disagree
8. D	tuder	ou think that an i	nternship or a coop	erative educati	on programme may h
8. D st ongly Agr	ree	ou think that an ints' careers in the	nternship or a coop e hospitality industr No Opinion	erative educati ry? Disagree	on programme may h Strongly Disagree
8. D st ongly Agr	ree Oo yo	ou think that an interest in the Agree ou think that a dispute think that an interest think that a dispute think tha	nternship or a coop e hospitality industr No Opinion	erative educations: Disagree ing of career ex	on programme may h Strongly Disagree
8. D st ongly Agr	ree Oo yo	ou think that an interest in the Agree ou think that a dispute think that an interest think that a dispute think tha	nternship or a coop e hospitality industr No Opinion ifferent understand	erative educations: Disagree ing of career ex	on programme may h

What qualifications do you have? (Please write the name of your qualification in

6.

10.		oyment at the hotel	•	manag	gement gra	duates during	uiei
		vidual attributes 1		lity to			
	perfe			J			
	Wor	k unit achieves	high levels of	f task			
	acco	mplishment					
	Wor	k effort relates	to a willingne	ess to			
	perfo	orm					
	Goo	d general knowled	ge of the hotel in	dustry			
	Wor	k performance effe	ective				
	Wor	ks effectively in te	am work				
	Othe	er (Please write)			1		
11.	•	ou consider a deg	•	an im	portant crit	terion when yo	u are
	select	ing a new employe	ee?				
Definitely		Probably	Uncertain	Unl	ikely	No	
12.		characteristic is th	e most important	for an	employee?		
		usiasm					
		ication					
		nmitment					
		ingness to learn					
		lworking					
	Dete	rmination					
	Spec	eific Skills (cus	tomer service	skills,			
	com	munication skills,	IT skills etc)				

Other (Please write)

Knowledge of the industry		
Personality		
Specific Skills (customer service skills,		
communication skills, IT skills etc)		
Qualifications		
Experience		
Other (Please write)		
	ortant to pos	sess as a
	ortant to pos	sess as a
n the hospitality industry? Knowledge of the industry Specific skills (customer service skills,	ortant to pos	sess as a
n the hospitality industry? Knowledge of the industry Specific skills (customer service skills, communication skills, IT skills etc).	ortant to pos	sess as a
Specific skills (customer service skills, communication skills, IT skills etc). Personality characteristics	ortant to pos	sess as a
n the hospitality industry? Knowledge of the industry Specific skills (customer service skills, communication skills, IT skills etc). Personality characteristics Qualifications	ortant to pos	sess as a
n the hospitality industry? Knowledge of the industry Specific skills (customer service skills, communication skills, IT skills etc). Personality characteristics Qualifications Experience	ortant to pos	sess as a
Knowledge of the industry Specific skills (customer service skills, communication skills, IT skills etc). Personality characteristics Qualifications Experience Attitude	ortant to pos	sess as a

What do you consider to be the most important criterion for a new employee?

13.

15.	Which						
	Com	munication skills	;				
	Usin	Using initiative skills					
	Hum	Human relations skills					
	Food						
	Prob	Problem solving skills					
	Self-	management skil	ls				
	Havi	ng multi-lingual	skills				
	Fron	t office skills					
	Othe	r (Please write)					
	prospe	ective employee?					
Strongly A		Agree	No Opinion	Disagree	Strongly Disagre		
Strongly A	Agree Do yo	Agree ou think a hospit	No Opinion	t degree would	Strongly Disagro		
	Agree Do you	Agree ou think a hospit	No Opinion	t degree would	d contribute more t		
17.	Do yo hotel l	Agree ou think a hospit business than hav Agree ou believe there	No Opinion tality management ring three years of No Opinion are any benefits r your organisation	degree would experience in to Disagree	d contribute more the hospitality indus		
17. Strongly	Do yo hotel l Agree Do yo manag	Agree ou think a hospit business than hav Agree ou believe there	No Opinion tality management ring three years of No Opinion are any benefits	degree would experience in to Disagree	d contribute more to the hospitality indus Strongly Disagre		
17. Strongly 2	Do yo manag	Agree ou think a hospir business than have Agree ou believe there gement degree for Agree	No Opinion tality management ring three years of No Opinion are any benefits r your organisation No Opinion	degree would experience in to Disagree for an employ n? Disagree Disagree	d contribute more to the hospitality industry Strongly Disagrates a hospitality Strongly Disagrates a hospitality should respect to the hospitality industry industry to the hospitality industry.		

20. Do you think that jobs in the hotel industry need a bachelor's degree in hospitality management for promotion purposes?

Strongly Agree	Agree	No Opinion	Disagree	Strongly Disagree

Thank you very much for taking the time to complete this questionnaire.

This survey will be valuable for developing students' abilities and for greatly improving the industry standards in New Zealand.

Appendix D – Primary Research Questions and Survey Questionnaire for managerial employees in the hotel industry

Primary Research Questions	Survey Questionnaire
1) What are the employees' working history,	4. How long have you been working
educational background, and experiences in	in the hospitality industry?
their career?	5. In which department did you start
	your career in the hospitality
	industry?
	6. What qualifications do you have?
2) What do they expect as graduate hospitality	10. What do you expect of hospitality
management students during their time of	management graduates during their
employment at the hotel?	employment at the hotel?
3) What are the human resource needs of the	12. What characteristic is the most
employers and the requirements they pose to	important for an employee?
potential employees? Where do employers see	13. What do you consider to be the
the point of entry of hospitality management	most important criterion for a new
graduates in the industry?	employee?
	14. Which of the following factors
	are the most important to possess as
	an employee in the hospitality
	industry?
	15. Which skill do you think was the
	most important for your career
	development?
4) What is the value of hospitality	11. Do you consider a degree
management degrees to employers?	qualification to be an important
	criterion when you are selecting a

	new employee?
	16. Do you think work experience is
	more important than a degree
	qualification for a prospective
	employee?
	17. Do you think a hospitality
	management degree will contribute
	more to the hotel business than
	having three years of experience in
	the hospitality industry?
	18. Do you believe there are any
	benefits to your organisation if an
	employee has a hospitality
	management degree?
	19. Do you think graduates with a
	bachelor's degree in hospitality
	should receive higher starting salaries
	than people who do not hold a
	degree?
	20. Do you think that jobs in the
	hospitality industry need a bachelor's
	degree in hospitality management for
	promotion purposes?
5) Do you recognise that there are some	7. Do you think that hospitality
different expectations or gaps between	management educators know the
hospitality management educators and	industry well?
employees in the hospitality industry? Do you	8. Do you think that an internship or
recognise that an internship or a cooperative	a cooperative education programme
education programme will help students'	may help students' careers in the
careers in the hospitality industry?	hospitality industry?
	9. Do you think that a different
	The second secon

understanding of career expectation
exists between educators and
managers in the hospitality industry?

Appendix E – Information sheet for survey

Information Sheet for Survey

- The hotel employees and employers
- Students at Auckland University of Technology



Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences

Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences



Dear Accor Hotel's employees

I am Ginny Kim and currently working in the Food and Beverage Department at Mercure Hotel Auckland. I have been working at the hotel for over three years. I am also a student and studying towards a Master of International Hospitality Management at AUT. This is the final year of my course and I am writing a thesis during the year. The title of the thesis is "Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences.

You are invited to participate in my survey. The aim of this study is to establish the perceived value of hospitality management or hotel management degrees amongst hospitality and hotel management undergraduate students and the hospitality industry. This study is also to identify the expectations and requirements differ between undergraduate hospitality and hotel management students, and employees in the hospitality industry. This survey is to be completed by approximately 300 students from AUT University, and approximately 150 employees from the Accor International Hotels especially Novotel, Mercure Hotel, and Ibis in New Zealand.

The reasons for inviting these particular students and employees to complete this questionnaire are:

Students - I want to focus on undergraduate students majoring in hospitality and hotel management at university for this study. This study does not include students from the Bachelor of Travel and Tourism and other institutions. However, Lincoln University has a Bachelor of Commerce and students can take a hotel institutional programme or major. Lincoln University does not offer a hospitality course. Therefore, I invite students from AUT University as the participants for this study.

Employees - the hotel industry is a huge industry and its employees are large workforces. Therefore, I have decided to select employees from a big worldwide international chain hotel in New Zealand. There are six brands under Accor hotels, which include Sofitel, Novotel, Mercure, All Seasons, Ibis and Formule 1. In New Zealand, there are approximately 23 hotels and resorts covering all prime business and holiday destinations. Therefore, I choose employees from the Accor International Hotels.

This study will provide significantly useful information for students, education providers, and the hospitality industry in New Zealand. The results of this survey may contribute to a considerable adjustment in the expectations between the students (and also education providers) and hospitality employees (the industry itself). Furthermore, the survey may contribute to developing students' abilities and greatly improving the industry standards in New Zealand.

It will take approximately 10 – 15 minutes to complete the questionnaire. Your participation in this study is completely voluntary. There are no foreseeable risks associated with this project. However, if you feel uncomfortable answering any questions, you can withdraw from the survey at any point. Your opinions are very important for my study. Your survey responses will be strictly confidential and data from this research will be reported only taken together. Your information will be coded and will remain confidential. The data will be stored in a locked cabinet in the School of Hospitality and Tourism, Auckland University of Technology, for 6 years. If you are interested in completing this questionnaire, please complete and return it using the attached self-addressed envelope. Completion of the attached questionnaire will be taken as indicating your consent to participate.

If you have questions at any time about the survey or the procedures, you may contact Ginny Kim at 021-3868 55 or email; junkimb3@aut.ac.nz and my supervisors Robert Steele or Tracy Harkison. Robert Steele, International Hospitality Management, AUT University, Wellesley Campus, phone 09 921-9999 ext. 8606 or email robert.steele@aut.ac.nz Tracy Harkison, International Hospitality Management, AUT

University, Wellesley Campus, phone 09 921- 9999 ext 8491 or email tracy.harkison@aut.ac.nz

This project has been reviewed and approved by the AUT University Ethics Committee on 09 July 2007. AUTEC Reference number is 07/59. If you have any concerns about the conduct of this research, please contact Ms Madeline Banda, Executive Secretary, AUT University Ethics Committee, telephone: 921 9999 ext 8044, e-mail: madeline.banda@aut.ac.nz. Postal Address: Private Bag 92006, Auckland 1142 or Room WA208, Level 2, WA Building, 55 Wellesley Street East, Auckland 1010.

Thank you very much for your participation.

Kind regards

Ginny Kim

Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences



Dear Hospitality Management students at Auckland University of Technology

I am studying towards a Master of International Hospitality Management. This is the second year of my course and I am doing a thesis. The title of the thesis is "Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences.

You are invited to participate in my survey. The aim of this study is to establish the perceived value of hospitality management or hotel management degrees amongst hospitality and hotel management undergraduate students and the hospitality industry. This study is also to identify the expectations and requirements differ between undergraduate hospitality and hotel management students and employees in the hospitality industry. This survey is to be completed by approximately 300 students from AUT University and approximately 150 employees from the Accor International Hotels especially Novotel, Mercure Hotel, and Ibis in New Zealand.

The reasons for inviting these particular students and employees to complete this questionnaire are:

Students - I want to focus on undergraduate students majoring in hospitality and hotel management at university for this study. This study does not include students from the Bachelor of Travel and Tourism and other institutions. However, Lincoln University has a Bachelor of Commerce and students can take a hotel institutional programme or major. Lincoln University does not offer a hospitality course. Therefore, I invite students from AUT University as the participants for this study.

Employees - the hotel industry is a huge industry and its employees are large workforces. Therefore, I have decided to select employees from a big worldwide international chain hotel in New Zealand. There are six brands under Accor hotels, which include Sofitel, Novotel, Mercure, All Seasons, Ibis and Formule 1. In New Zealand, there are approximately 23 hotels and resorts covering all prime business and holiday destinations. Therefore, I choose employees from the Accor International Hotels.

This study will be significantly useful information for students, education providers, and the hospitality industry in New Zealand. The results of this survey may contribute to a considerable adjustment in the expectations between the students (and also education providers) and hospitality employees (the industry itself). Furthermore, this survey may contribute to developing students' abilities and greatly improving the industry standards in New Zealand.

It will take approximately 10 – 15 minutes to complete the questionnaire. Your participation in this study is completely voluntary. There are no foreseeable risks associated with this project. However, if you feel uncomfortable answering any questions, you can withdraw from the survey at any point. It is very important for my study to know your opinions. Your survey responses will be strictly confidential and data from this research will be reported only in the aggregate. Your information will be coded and will remain confidential. The data will be stored in a locked cabinet in the School of Hospitality and Tourism, Auckland University of Technology, for 6 years. If you are interested in participating in this questionnaire, please complete it. Completion of the attached questionnaire will be taken as an indication of your consent to participate.

If you have questions at any time about the survey or the procedures, you may contact Ginny Kim at 021-3868 55 or email; junkimb3@aut.ac.nz and my supervisors Robert Steele or Tracy Harkison. Robert Steele, International Hospitality Management, AUT University, Wellesley Campus, phone 09 921-9999 ext. 8606 or email robert.steele@aut.ac.nz Tracy Harkison, International Hospitality Management, AUT

University, Wellesley Campus, phone 09 921- 9999 ext 8491 or email tracy.harkison@aut.ac.nz

This project has been reviewed and approved by the Auckland University of Technology Ethics Committee on 09 July 2007. AUTEC Reference number is 07/59. If you have any concerns about the conduct of this research, please contact Ms Madeline Banda, Executive Secretary, AUT University Ethics Committee, telephone: 921 9999 ext 8044, e-mail: madeline.banda@aut.ac.nz. Postal Address: Private Bag 92006, Auckland 1142 or Room WA208, Level 2, WA Building, 55 Wellesley Street East, Auckland 1010.

Thank you very much for your participation.

Kind regards

Ginny Kim

Appendix F – Ethics approval letter from AUTEC



MEMORANDUM

Auckland University of Technology Ethics Committee (AUTEC)

To: Robert Steele

From: Madeline Banda Executive Secretary, AUTEC

Date: 10 August 2007

Subject: Ethics Application Number 07/59 Career expectations and requirements

of undergraduate hospitality students and the hospitality industry: an

analysis of differences.

Dear Robert

Thank you for providing written evidence as requested. I am pleased to advise that it satisfies the points raised by the Auckland University of Technology Ethics Committee (AUTEC) at their meeting on 14 May 2007 and that on 6 August 2007, as the Executive Secretary of AUTEC I approved your ethics application at their meeting. This delegated approval is made in accordance with section 5.3.2.3 of AUTEC's *Applying for Ethics Approval: Guidelines and Procedures* and is subject to endorsement at AUTEC's meeting on 10 September 2007.

Your ethics application is approved for a period of three years until 6 August 2010.

I advise that as part of the ethics approval process, you are required to submit to AUTEC the following:

A brief annual progress report indicating compliance with the ethical approval given using form EA2, which is available online through http://www.aut.ac.nz/about/ethics, including when necessary a request for extension of the approval one month prior to its expiry on 6 August 2010;

• A brief report on the status of the project using form EA3, which is available online

through http://www.aut.ac.nz/about/ethics. This report is to be submitted either

when the approval expires on 6 August 2010 or on completion of the project,

whichever comes sooner;

It is also a condition of approval that AUTEC is notified of any adverse events or if the

research does not commence and that AUTEC approval is sought for any alteration to the

research, including any alteration of or addition to the participant documents involved.

You are reminded that, as applicant, you are responsible for ensuring that any research

undertaken under this approval is carried out within the parameters approved for your

application. Any change to the research outside the parameters of this approval must be

submitted to AUTEC for approval before that change is implemented.

Please note that AUTEC grants ethical approval only. If you require management approval

from an institution or organisation for your research, then you will need to make the

arrangements necessary to obtain this.

To enable us to provide you with efficient service, we ask that you use the application

number and study title in all written and verbal correspondence with us. Should you have

any further enquiries regarding this matter, you are welcome to contact Charles Grinter,

Ethics Coordinator, by email at charles.grinter@aut.ac.nz or by telephone on 921 9999 at

extension 8860.

On behalf of the Committee and myself, I wish you success with your research and look

forward to reading about it in your reports.

Yours sincerely

Madeline Banda

Executive Secretary

Auckland University of Technology Ethics Committee

Cc: Ginny Kim junkimb3@aut.ac.nz

138

Appendix G – Approval survey for employees

Dear A. Hotels' Human Resource Managers

I am Ginny Kim and working in the Food and Beverage Department at M. Hotel Auckland. I have been working at the hotel for over three years. I am also a student and studying towards a Master of International Hospitality Management at AUT. This is the final year of my course and I am writing a thesis during this year. The title of the thesis is "Career expectations and requirements of undergraduate hospitality students and the hospitality industry: An analysis of differences.

The aim of this study is to establish the perceived value of hospitality management degrees amongst hospitality management undergraduate students and the hospitality industry. This study also hopes to identify how the career expectations and requirements differ between undergraduate hospitality management students, and employees in the hospitality industry.

I am going to start survey for my study. This survey is to be completed by students at AUT University. The survey also includes employees in the hotel industry. Therefore, I would like to invite Accor Hotels' employees (especially the positions ranging from middle management to top management levels) to participate in my survey. I need your help to complete my study. I would like to request a favour to you, that once I have sent the survey forms, which you will give them to your employees. I will enclose a self-addressed envelope in each survey form for the returning of the survey.

It will take approximately 10 minutes to complete the questionnaire. Employees' participation in this study is completely voluntary. There are no foreseeable risks associated with this project. Employees' opinions are very important for my study. Employees' survey responses will be strictly confidential. All data will be coded thus will remain confidential.

I am attaching my manager's (Stuart Cox) reference letter. Please refer to his letter.

I would like to send you survey forms, after receiving your reply. Furthermore, please let me know how many employees (from department supervisors to general manager) in your hotel. Thank you very much. I look forward to hearing from you as soon as possible.

Regards,

Ginny Kim

Appendix H – Reference letter

Stuart Cox

Food & Beverage Manager

Mxxxx Hotel Auckland

8 Customs Street

Auckland

Dear Colleague

I am writing this letter in support of Ginny Kim's survey she is undertaking as research for

AUT University.

She would be very grateful if you could find the time to assist in this worthwhile survey

and I personally support her in this research as the findings may benefit the industry as a

whole and provide some insight into the motivations and expectations of those who choose

to study hospitality.

Please do not hesitate to contact Ginny directly by e-mailing junkimb3@aut.ac.nz should

you have any questions in regards to this survey.

Kind Regards

(Not signed due to electronic transmission)

Stuart Cox

Food & Beverage Manager

Mxxxx Hotel Auckland

(Not signed due to electronic transmission)

141

Appendix I – Data modification schedule

Some data merged contexts of answers logically because some questions had 14 examples of responses so it was difficult to obtain accurate results of data analysis. Therefore, the researcher used a transform function (record into different variables) of SPSS to merge samples of responses in each question. Data was analysed by methods of Crosstabulation and Frequencies in the Descriptive Statistics.

Table 1 Data Modify Schedule

Data Modif	y Schedule	
Question	Original contexts of answers	Change
E : Q1	The six hotels merged to four	Four regions hotels:
	regions hotels.	1. Auckland region = 1+2
	Six hotels:	2. Rotorua region = 3
		3. Wellington region = 4
	Auckland Hotel A	4. South Island = 5+6
	2. Auckland Hotel B	
	3. Rotorua Hotel C	
	4. Wellington Hotel D,	
	5. Queenstown Hotel E	
	6. Hotel Christchurch F	
S : Q3, Q4,	Merged 5-point scales to 3-point	3-point scales:
	scales.	1. Certain = 1+2
E : Q7, Q11		2 Uncertain = 3
	5-point scales:	3 No = 4+5
	1 Definitely	
	2 Probable	
	3 Uncertain	
	4 Unlikely	
	5 No	

S ; Q5, Q8,	Merged 5-point scales to 3-point	3-point scales:
Q9, Q10,	scales	1 Agree = 1+2
Q11, Q18,		2 No Opinion = 3
Q19, Q20,	5-point scales:	3 Disagree =4+5
Q21, Q22	1 Strongly Agree	
	2 Agree	
E : Q8,Q9,	3 No Opinion	
Q16, Q17,	4 Disagree	
Q18, Q19,	5 Strongly Disagree	
Q20		
S : Q6	Merged 7 multiple choices to 5	4 multiple choices:
		1 Food and Beverage and
	7 multiple choices:	Conference = 1+7
	1 Food and Beverage	2 House Keeping=2
	2 House Keeping	3 Front Office= 3
	3 Front Office	4 Administer & General =4+5+6+7
	4 Sales and Marketing	
	5 Finance	
	6 Human Resources	
	7 Other	
S: Q12	Merged 8 multiple choices to 4	4 Multiple choices:
	multiple choices.	
E: Q12	8 multiple choices:	1 Enthusiasm = 1
	1 Enthusiasm	2 Commitment = 2+3+5+8
	2 Dedication	3 Willing to learn = 4
	3 Commitment	4 Specific Skills = 7
	4 Willing to learn	
	5 Hardworking	
	6 Determination	
	7 Specific Skills	
	8 Other.	

S: Q13	Merged 6 multiple choices to 4	4 multiple choices:
	multiple choices	1 Knowledge of the industry and
E: Q13		Experience = 1+5
	6 multiple choices:	2 Personality =2
	1 Knowledge of the industry	3 Specific Skills =3
	2 Personality	4 Qualifications = 4+6
	3 Specific Skills	
	4 Qualifications	
	5 Experience	
	6 Other	
S: Q14	Merged 8 multiple choices to 4	4 multiple choices :
	multiple choices	1) Knowledge of the Industry and
E: Q14		Experience = 1+5
	8 multiple choices:	2) Personality = 2+6+7
	1 Knowledge of the industry,	3) Specific Skills = 3+8
	2 Specific Skills	4) Qualifications = 4
	3 Personality	
	4 Qualifications	
	5 Experience	
	6 Attitude	
	7 Commitment	
	8 Other	
S: Q16	Merged 6 choices to 4 choices	4 Choices:
		1 Low Management = 1
	6 Choices:	2 Mid Management = 2
	1 Department Supervisor	3 Top Management = 3+4
	2 Department Manager	4 Self-employee = 5
	3 Senior Manager	
	4 Top Management	
	5 Self-employee	
	6 Other	

S: Q17	Merged 5 choices to 3 choices	3 Choices:
		1 Mid-Management = 1
	5 Choices:	2 Top Management = 2+3
	1 Department Management	3 Self-employ = 4
	2 Senior Management	
	3 Top Management	
	4 Self-employ	
	5 Other	
G. 015	Manual O multiple shakes to 5	5
S: Q15	Merged 9 multiple choices to 5	5 multiple choices:
	multiple choices	1 Communication skills = 1
E: Q15		2 Using initiative skills = $2+7+8$
	9 multiple choices:	3 Human relation skills = 3
	1 Communication skills	4 Problem solving skills = 4
	2 Using initiative skills	5 Self-management skills = 5
	3 Human relations skills	
	4 Food and Beverage skills	
	5 Problem solving skills	
	6 Self-management skills	
	7 To be multi-lingual skills	
	8 Front Office skills	
	9 Other	

E: Q3	Merged 14 different multiple	5 different multiple choices:
	choices to 5 different multiple	1 FO & Reservation = 1+6+7
	choices.	2 F&B = 3+4+5
		3 Executive Office = 12
	14 different multiple choices:	4 House Keeping = 8
	1 Reservations	5 Administer & General =
	2 Sales and Marketing	2+9+10+11+13
	3 Banquet Operation or	
	Conferences	
	4 Food & Beverage	
	5 Kitchens	
	6 Rooms and Front Desk	
	7 Guest Services (Door person,	
	Concierge, Telephones)	
	8 Housekeeping	
	9Purchasing and Stores	
	10 Engineering	
	11 Human Resources	
	12 Executive Office	
	13 Finance and Accounts	
	14 Other	
E: Q4	merged 10 different multiple	3 different multiple choices:
	choices to 3 different multiple	1 less than 5 years = $1+2+3$
	choices	2 5-10 years =4+5
		3 11+ years =6+7+8+9+10
	9 different multiple choices:	
	1 Less than 1 year	
	2 1-2 years	
	3 2-4 years	
	4 5-7 years	
	5 8-10 years	
	6 11-15 years	

	7 16-20 years	
	8 21-25 years	
	9 26-10 years	
	10 More than 30 years	
E: Q5	Merged 11 different choices to 5	5 different multiple choices :
	different choices.	1 FO & Reservation = 3+9+11
		2 F&B = 1+7+11
	11 different multiple choices:	3 Executive Office = 11
	1 Food and Beverage	4 House Keeping = 2
	2 Housekeeping	5 Administer & General =
	3 Room and Front Office	4+5+6+8+10
	4Sales and Marketing	
	5 Finance, Accounts, and Credit	
	6 Human Resources	
	7 Kitchens	
	8 Engineering	
	9 Guest Services (Door Person,	
	Concierge, Telephones)	
	10 Purchasing and Stores	
	11 Other	
E: Q6	Merged 10 different multiple	5 different multiple choices:
	choices to 5 different multiple	1 Secondary =1
	choices.	2 Certificate & Diploma
	10 different multiple choices:	=2+3+4+5+6+8
	1 Secondary School Certificate	3 Degree = 7
	2 Vocational Qualification (Trade,	4 Postgraduate = 9
	New Zealand Certificate)	5 No qualification = 10
	3 Polytechnic Certificate	
	4 Polytechnic Diploma	
	5 University Certificate	
	6 University Diploma	

7 University Bachelor Degree	
8 Private School Qualification	
9 Postgraduate Qualification	
10 Other	

Appendix J - Glossary for this study

The following definitions are provided to help with understanding of terms used in this study.

Hospitality

An early definition of hospitality included any and all businesses and services whose primary objective was serving people outside of a private home (Barrows & Bosselman, 1999). Cassee and Reuland (1983) as cited in a study by Brotherton (1999) wrote that a definition of hospitality is a harmonious mixture of food, beverage, and/or shelter, a physical environment and the behaviour and attitude of people.

Hospitality industry

Businesses operate to meet lodging, food and beverage, vacation business, and recreational business needs and wants of customers. The industry includes accommodation businesses, restaurants, bars, casinos, catering, resorts, clubs and any businesses that offer food or shelter for profit to people away from home (Buergemeister, 1983).

Hospitality management degree

A bachelor's degree (normally obtained after form 6 or 7 of college or university-level academic pursuit) that has as its curriculum focus a field of multidisciplinary study which brings the perspectives of many disciplines, especially the training and knowledge required for future employment in the hospitality industry (Ricci, 2005). It tends to focus on preparing students, usually for management positions in the hospitality industry.

International hotel chain

A hotel chain is a collection or grouping of hotels under one recognizable brand operated by a management company. An International hotel chain claims to be the largest hotel chain in the world, in terms of the number of properties. For example, Accor Hotels, Best Western International and Star Wood Hotels.

<u>Internship</u>

An internship is when a person works in a temporary position with an emphasis on job training, rather than just employment. It is similar to an apprenticeship. Usually undergraduate hospitality students have an internship programme during their course. This provides opportunities for students to gain work experience in the hospitality industry (Lee, 2006).

Back of the house

The back of the house supports and services areas not usually seen by guests of a hotel such as kitchen staff, human resources staff, finance staff, and housekeeping staff. The major responsibility of the back of the house is the quality of the products or service offered to customers (Powers & Barrows, 2006).

Front of the house

Front of the house employees directly support and service customers, such as front desk staff and restaurant staff. The most important responsibility of the front of the house is guest satisfaction, with a particular emphasis on personal service (Powers & Barrows, 2006).

Cooperative education programme (CEP)

A cooperative education programme is part of an academic course that allows students to apply classroom theory in practical work settings and gain personal, academic, work skills, and work competencies (University of Central Florida, 2007). Cooperative education is a partnership among students, education providers and employers, with specified responsibilities for each party (Lee, 2006). In New Zealand, undergraduate hospitality students need to complete this as part of the normal academic curriculum while integrating a number of work terms into their programme (Harkison, 2004a).

Appendix K – Number of employees by RTO Commercial Accommodation Monitor)

					Last	Last 13 Months							
Regional Tourism Organisation (RTO)	Dec 06	Jan 07	Feb 07	Mar 07	Apr 07	May 07	Jun 07	70 Inc	Aug 07	Sep 07	Oct 07	Nov 07	Dec 07
	2 24 0 224	4 262 007			. 200 010 0		. 000 222	0.00 0.00	000 000		430 000		700 600
Ordal Ordal	3,210,231	4,505,007	-		2,013,960 1,941,503		1,009,222	20,203,310	20,051,2	_	660,004,7		770,007,0
Iorthland KTO	482 000	563,742	205,402	183,851	143,542	380,580	276.133	416.055	417,698	431 013	456 218	543.051	486 350
oromandel RTO	93,960	157,369	92,269	73,268	59,259	28,697	24,921	24,073	25,584	29,383	38,460	46,163	88,101
Valkato RTO	91,551	112,440	98,210	119,987	96,806	75,923	80,263	83,949	79,009	86,580	84,264	83,759	84,628
lay of Plenty RTO	147,456	259,680	122,964	114,042	109,378	72,284	68,559	70,115	59,780	78,317	83,843	83,574	153,709
astland RTO	56, 109	102,574	43,800	39,717	36,027	23,983	21,461	21,513	17,910	23,854	29,646	28,070	66,489
otorua RTO	181,072	222,786	185,006	184,441	169,173	118,658	107,919	139,177	113,470	143,902	151,266	165,042	175,419
ake Taupo RTO	98,877	135,633	107,964	103,418	90,687	56,874	56,091	76,189	65,751	77,618	75,108	81,706	91,747
uapehu RTO	26,253	30,835		29,405	22,846	12,552	16,171	53,414	51,260	53,892	26,012	23,735	23,840
aranaki RTO	49,254	79,468	56,011	56,883	48,284	33,227	35,675	36,225	30,409	36,825	45,139	45,210	53,301
awkes Bay RTO	113,718	160,258	103,874	109,838	89,077	56,487	54,545	58,703	269'65	64,360	81,812	75,577	111,080
lanawatu/Wanganui RTO	20,181	24,963		19,794	16,807	13,153	12,289	12,214	13,611	14,783	15,194	16,352	19,292
Vairarapa RTO	23,799	31,675		24,026	19,335	12,989	12,921	12,663	10,988	15,006	15,831	15,999	21,086
Wellington RTO	143,331	163,076	182,850	188,576	156,288	138,322	132,535	142,228	135,792	155,794	152,375	175,300	158,614
1arlborough RTO	80,226	102,100	89,024	84,319	67,126	37,758	36,177	34,880	35,287	40,978	48,481	62,370	79,524
lelson/Tasman RTO	152,277	269,739	166,065	135,342	102,449	61,888	49,300	51,701	54,218	54,150	70,307	93,888	157,818
Vest Coast RTO	135,160	178,031	173,660	150,877	117,196	68,375	55,839	64,361	59,280	75,428	92,888	118,420	135,577
anterbury RTO	416,585	501,409	461,095	441,868	387,155	259,918	244,004	305,362	286,833	311,235	355,455	409,492	428,797
ackenzie/Ctrl SI RTO	72,545	88,572	76,808	69,698	54,204	30,415	25,771	29,196	32,677	38,439	46,000	59,134	71,470
Queenstown/Crtl Otago/Wanaka RTO	309,997	428,936	335,879	316,466	270,508	158,309	151,173	306,088	303,678	246,808	211,049	254,183	309,781
Dunedin RTO	90,709	110,262	107,915	114,288	94,480	70,178	55,481	60,050	60,414	67,360	76,512	83,275	88,796
outhland/Fiordland RTO	93,635	117,927	122,221	118,691	82,996	48,297	35,213	36,815	37,397	45,352	60,471	86,552	93,673
Other: Wanganui	16,761	23,230	22,947	18,810	14,373	12,401	12,022	12,573	13,913	14,110	13,482	15,494	20,496
Ather: Palmerston North	27,164	32,019	41,581	39,712	31,338	31,404	28,406	32,062	42,124	36,784	31,981	32,921	27,818
ther: Wgtn/Kapiti/Wairarapa 211,337	na 211,337	249,354	263,823	266,849	217,299	182,123	172,629	185,699	178,156	203,704	206,033	230,954	221,751
Other: Centre Stage	443,840	621,194	518,911	486,510	386,874	281,768	258,106	272,279	267,661	298,832	324,822	387,212	459,093
Other: Ashburton	14,466	14,021	15,051	11,940	11,328	8,392	13,200	20,822	22,248	20,347	15,880	11,765	15,401
Other: Waitaki	33,898	49,500	39,917	35,877	30,666	17,597	13,509	15,566	16,562	18,078	23,448	26,293	34,338